

The World's Largest Open Access Agricultural & Applied Economics Digital Library

# This document is discoverable and free to researchers across the globe due to the work of AgEcon Search.

Help ensure our sustainability.

Give to AgEcon Search

AgEcon Search http://ageconsearch.umn.edu aesearch@umn.edu

Papers downloaded from **AgEcon Search** may be used for non-commercial purposes and personal study only. No other use, including posting to another Internet site, is permitted without permission from the copyright owner (not AgEcon Search), or as allowed under the provisions of Fair Use, U.S. Copyright Act, Title 17 U.S.C. Agricultural Shocks and Conflict in the Short- and Long-Term: Evidence from Desert Locust Swarms

**Pierre Biscaye** 

University of California at Berkeley

pbiscaye@berkeley.edu

Selected Paper prepared for presentation at the 2023 Agricultural & Applied Economics Association Annual Meeting, Washington DC; July 23-25, 2023

Copyright 2023 by Pierre Biscaye. All rights reserved. Readers may make verbatim copies of this document for non-commercial purposes by any means, provided that this copyright notice appears on all such copies.

# Agricultural Shocks and Conflict in the Short- and Long-Term: Evidence from Desert Locust Swarms

Pierre E. Biscaye<sup>†</sup>

This version: May 15, 2023 Latest version available here. Preliminary; please do not cite.

#### Abstract

This paper tests the importance of changes in opportunity costs related to agriculture on the risk of violent conflict using data on locust swarms and conflict collapsed to annual  $0.25^{\circ}$  (approx. 28km<sup>2</sup>) grid cell observations across Africa and the Arabian peninsula. The identification exploits exogenous local variation in locust swarm exposure driven by patterns in swarm movements together with weather controls and grid cell and country-by-year fixed effects to identify causal impacts of these agricultural shocks. Locust swarms decrease the likelihood of violent conflict event in a given year by around 20%. Effects are driven by areas with crop and pasture land, and there is no evidence of conflict spillovers to nearby areas. The impacts are largest for swarms that arrive in the off-season or planting season for major crops, based on national crop calendars, and the patterns are not consistent with effects on conflict driven by changes in conflict opportunity costs related to agriculture. This points to the availability of non-agricultural livelihood opportunities and to alternative factors such as psychological impacts and relief efforts less often discussed in this literature as crucial in determining whether an agricultural shock increases conflict risk. In contrast to short term negative effects on conflict, cells affected by the 2003-2005 major desert locust upsurge were 62% more likely to experience any conflict in a given year afterward. Absolute impacts are increasing over time alongside a general increase in conflict in the sample countries, suggesting affected areas are made vulnerable to future shocks which precipitate conflict.

**JEL codes:** Q54; D7; Q10; O13; N57

Keywords: desert locusts; natural disasters; conflict; agriculture; Africa

I thank Daniel Agness, Maximillian Auffhammer, Alain de Janvry, Joel Ferguson, Ethan Kapstein, Ethan Ligon, Jeremy Magruder, Ted Miguel, Anouk Rigterink, Betty Sadoulet, Jed Silver, and seminar participants at UC Berkeley, Princeton University, the Empirical Studies of Conflict Conference, the Pacific Conference for Development Economics, and the Midwest International Economic Development Conference for helpful comments. Luka Marcel, Jane McLoughlin, and Aiko Sudijono provided excellent research assistance. All errors are my own.

<sup>&</sup>lt;sup>†</sup> Department of Agricultural and Resource Economics, University of California at Berkeley; pbiscaye@berkeley.edu.

## 1 Introduction

A growing literature explores the role of agricultural shocks in increasing the risk of conflict (e.g., Dube and Vargas 2013; Harari and La Ferrara 2018; Maystadt and Ecker 2014; McGuirk and Nunn 2021; Miguel, Satyanath, and Sergenti 2004), with particular attention to Africa where the number of conflict events has been increasing rapidly since the late 2000s. Understanding this relationship is increasingly important in the context of climate change as the frequency and severity of weather-related shocks increases. This paper provides new evidence on the relationship between agricultural shocks and conflict in the short- and longterm by analyzing outbreaks of desert locusts, the world's most dangerous and destructive migratory pest (Cressman, Van der Elstraeten, and Pedrick 2016; Lazar et al. 2016).

Using data on the location and timing of desert locust swarm observations from the Food and Agricultural Organization of the United Nations and of conflict events from the Armed Conflict Location & Event Data Project, I estimate a model of conflict at the annual level for 0.25° (around 28 km<sup>2</sup>) grid cells between 1997-2018 across Africa and the Arabian peninsula. The identification exploits exogenous local variation in locust swarm exposure driven by swarm flight patterns to identify causal impacts of these shocks. The regressions control for current and lagged local weather realizations, country-by-year fixed effects, and cell fixed effects, generally following the approach in Burke, Hsiang, and Miguel (2015) and Harari and La Ferrara (2018) and other recent papers.

As in these papers and others in the climate and conflict literature, I find that deviations in rainfall and temperature from historical norms increase conflict risk in the short term. On the other hand, the presence of a locust swarm in a location decreases the likelihood of any violent conflict event occurring in the same year, by around 20% in the most conservative subsamples. This result contrasts with most of the literature on climate and conflict (Burke, Hsiang, and Miguel (2015)) but is robust to a variety of alternative specifications and subsamples. Negative impacts on conflict risk persist into the following year. Impacts of swarms on conflict are largely local with no significant spillovers into surrounding areas up to 500km away, indicating effects are not driven by displacement of conflict to nearby areas.

The negative average impact of swarms on conflict at the annual level is driven largely by swarms that arrive in the local off-season or planting season for major crops. Swarms arriving during the growing and harvest period have no statistically significant effect. The pattern of results is similar across regions with very different crop calendars. The literature on agricultural shocks and conflict typically interprets the relationship through the lens of a simple model of opportunity costs related to agriculture and returns to fighting over agricultural output, but the results for locust swarms are not consistent with such a model. This implies an important role for opportunity costs related to non-agricultural work, and for other factors less frequently discussed in this literature such as psychological impacts, social cohesion, and relief efforts.

I next explore long-term impacts of an agricultural shock on conflict using an event study of the 2003-2005 major desert locust upsurge. While contemporaneous effects of locust swarms on conflict are negative, cells that experienced a locust swarm during this outbreak are 62% more likely to experience any conflict in a given year afterward relative to cells that were not affected by the upsurge. The impact on conflict risk increases rather than decreases over time, suggesting affected areas do not recover from this agricultural disaster over time but rather remain vulnerable to future shocks which may precipitate conflict.

This paper makes three main contributions. First, I provide new evidence on the drivers of conflict, particularly in Africa (see e.g., Blattman and Miguel 2010; Collier and Hoeffler 1998; McGuirk and Burke 2020; Miguel, Satyanath, and Sergenti 2004), testing the most commonly discussed mechanisms linking agricultural shocks to conflict. Using local variation in desert locust swarms, a novel agricultural shock in this literature, I find that an extreme agricultural shock decreases the short-term risk of conflict, in contrast to much of the empirical evidence (see Burke, Hsiang, and Miguel (2015) for a review). Despite local variation in locust swarm destruction, I find no evidence of conflict displacement or spillovers. I build on recent papers analyzing the role of agricultural seasonaliy in conflict (Crost et al. 2018; Guardado and

Pennings 2021; Ubilava, Hastings, and Atalay 2022, and show that the impacts of locust swarms on conflict by swarm timing do not align with seasonal differences in effects of locust on agricultural productivity. How an agricultural shock affects conflict risk is therefore not necessarily a simple function of the impact on agricultural production.

Second, I add to our understanding of the economic and social effects of natural disasters (see Botzen, Deschenes, and Sanders (2019) and Klomp and Valckx (2014) for reviews) as one of the first to study long-term impacts on the risk of violent conflict. This is a broad literature but the evidence on long-term impacts of disasters such as hurricanes and droughts is limited, inconclusive, and focused on a small number of outcomes (Botzen, Deschenes, and Sanders 2019; Cavallo et al. 2013; Gignoux and Menéndez 2016; Heger and Neumayer 2019; Hsiang and Jina 2014; Kocornik-Mina et al. 2020). I show that desert locust swarms—a natural disaster that operates almost entirely through effects on the agricultural sector—increase conflict risk over the following 15 years.

Third, I expand the evidence base on the economic impacts of agricultural pest shocks (Bradshaw et al. 2016; Oerke 2006). A large literature reports on the short-term impacts of agricultural pests on agricultural production, household consumption, or coping mechanisms, but few studies consider broader or long-term impacts (some exceptions include Baker, Blanchette, and Eriksson (2020), Banerjee et al. (2010), Conte, Tapsoba, and Piemontese (2021), and De Vreyer, Guilbert, and Mesple-Somps (2015)). The range of many agricultural pests is expanding due to climate change and globalization, and though locust outbreaks have become less frequent in recent decades desert locusts are ideally situated to benefit from climate change (ASU 2020). Policies to address this challenge should be informed by estimates of the costs outside of immediate agricultural losses. This paper analyzes how destruction caused by an important migratory pest affects local risk of conflict in the short and long term across Africa and the Arabian peninsula.

The remainder of the paper is organized as follows. Section 2 provides background on desert locusts and discusses how agricultural shocks may affect the risk of violent conflict.

Section 3 discusses the data used in the analyses and Section 4 outlines the empirical approaches. Section 5 shows and discusses the results on short-term impacts of swarms on conflict, spatial spillovers, agricultural destruction and seasonality, and long-term impacts. Section 6 concludes.

## 2 Background

#### 2.1 Desert locusts

Damages from desert locust swarms—the world's most dangerous and destructive migratory pest (Cressman, Van der Elstraeten, and Pedrick 2016; Lazar et al. 2016)—can be extreme.During the locust upsurge in 2003-2005 in North and West Africa, 100, 90, and 85% losses on cereals, legumes, and pastures respectively were recorded, affecting more than 8 million people (Renier et al. 2015; Brader et al. 2006). In the most recent upsurge from 2019-2021 in East Africa and the Arabian Peninsula, over 40 million people in 10 countries faced severe food insecurity due to crop destruction (Food and Agriculture Organization of the United Nations (FAO) 2022a). The food insecurity motivates large numbers of individuals to move away from locust-affected areas: the World Bank estimates that 8 million people were internally displaced during this most recent upsurge (The World Bank 2020).

Small numbers of locusts are always present in desert 'recession' areas from Mauritania to India, posing little threat to livelihoods.<sup>1</sup> But favorable climate conditions—periods of repeated rainfall and vegetation growth overlapping with the breeding cycle—can lead to exponential population growth. Unique among grasshopper species, after reaching a particular population density desert locusts undergo a process of 'gregarization' wherein they mature physically and form large bands or swarms (Symmons and Cressman 2001). This process can lead to 'outbreaks' or 'upsurges'where locust swarms spread from their

<sup>&</sup>lt;sup>1</sup>Additional detail on desert locusts is included in Appendix B. Any time I use 'locusts' in this paper I am referring exclusively to desert locusts.

desert breeding areas. Few locust swarms are observed outside of major outbreaks.

Figure 1 displays the locations of desert locust swarm observations in the FAO Locust Watch database by year for the sample countries and years for this analysis. As illustrated by the figure, locust swarms are not observed with any regularity over time or space. The countries affected by the 2003-2005 upsurge are not the same as those that have experienced more recent outbreaks. Desert locusts are migratory, moving on after consuming all available vegetation, rather than becoming endemic. The arrival of a swarm is thus a locally and temporally concentrated natural disaster where all crops and pastureland are at risk (Hardeweg 2001) but does not signal a permanent change in local agricultual pest risk.

#### [Figure 1 here]

Locust outbreaks end due to a combination of migration to unfavorable habitats, failure of seasonal rains, and control operations (Symmons and Cressman 2001). Farmers have no proven effective recourse when faced with the arrival of a locust swarm(Dobson 2001; Hardeweg 2001). The only current viable method of swarm control is direct spraying with pesticides (Cressman and Ferrand 2021, which can take days to have effects as well as being slow and costly to organize and requiring robust locust monitoring infrastructure.

Locust swarms vary in density and extent (Symmons and Cressman 2001). The average swarm includes around 50 locusts per  $m^2$  with a range from 20-150, and can cover under  $1km^2$  to several hundred. Swarms fly for 9-10 hours each day, from a few hours after sunrise to an hour or so before sunset. They fly downwind and can easily move 100km or more in a day even with minimal wind. As a result of this flight pattern, some areas in the flight path of a locust swarm are spared any agricultural destruction. This can be seen in Figure 1 by many unaffected areas even in countries with large numbers of swarms during the 2003-2005 upsurge. Where swarms land during an outbreak is determined largely by patterns of wind direction and speed over time from the initial breeding areas. I leverage this quasi-random variation in the areas affected by swarms to identify their impact on conflict.

#### 2.2 Agricultural shocks and conflict

While conflict between states has become less common in recent decades, civil conflicts are increasing in frequency in many parts of the world. These conflicts have a variety of proximate causes, but a growing literature explores the impacts of climate or weather on conflict (see Burke, Hsiang, and Miguel (2015) and Dell, Jones, and Olken (2012) for reviews), generally finding that deviations from historical norms increase conflict risk.

Though some studies have pointed to psychological or infrastructural effects of weather shocks in explaining impacts on conflict (Chemin, De Laat, and Haushofer 2013; Hsiang and Burke 2014; Sarsons 2015; Witsenburg and Adano 2009), the majority of papers focus on the role of changes in agricultural productivity and opportunity costs of conflict, in line with the Chassang, Miquel, et al. (2009) model. Many studies find support for the argument that impacts of agricultural shocks on the decision to engage in conflict are driven by changes in the opportunity cost of fighting (Crost et al. 2018; Fjelde 2015; Guardado and Pennings 2021; Harari and La Ferrara 2018). Others emphasize changes in the potential returns to conflict over outputs following an agricultural shock (McGuirk and Nunn 2021; Ubilava, Hastings, and Atalay 2022). Both mechanisms are common to the literature on the sources of conflict more generally (see e.g., Blattman and Miguel 2010; Chassang, Miquel, et al. 2009; Collier and Hoeffler 1998, 2004; Dal Bó and Powell 2009; Dal Bó and Dal Bó 2011; Dube and Vargas 2013; Fearon 1995; McGuirk and Burke 2020).

In the parts of the world in the migratory range of desert locusts, a large share of households rely on agriculture for their livelihoods, making agricultural productivity shocks particularly important for conflict. Consider a simple Roy model in which individuals choose their occupation from between agricultural production, non-agricultural work, and fighting to capture resources.<sup>2</sup> Adverse agricultural shocks reduce the opportunity cost of fighting by reducing the expected gains from engaging in agricultural production. At the same time, reduced agricultural productivity reduces the potential gains from fighting to the extent that

<sup>&</sup>lt;sup>2</sup>This simple model is developed in more detail in Appendix C.

agricultural outputs are an important share of the resources that could be captures. Nonagricultural livelihood opportunities play an important role in setting a lower bound on how far the opportunity cost of fighting falls following a negative agricultural shock, assuming returns to non-agricultural work are less affected by such shocks.

Agricultural shocks may also have long-term impacts on conflict risk, but the evidence is limited. Crost et al. (2018) and Harari and La Ferrara (2018) find impacts of agricultural production shocks persist in the short-term but do not consider impacts beyond two years. To my knowledge, only Iyigun, Nunn, and Qian (2017) consider how an agricultural shock impacts conflict in the long term, though they stand out in studying impacts of a permanent increase to agricultural productivity. They find that introducing potatoes to Europe, the Near East, and North Africa led to a large and permanent reduction in the risk of conflict in subsequent centuries. Adverse agricultural shocks may similarly affect long-term outcomes if they cause persistent decreases in agricultural productivity, such as through destruction of infrastructure, environmental degradation, or reduction of household productive assets. Lasting reductions in agricultural productivity could increase the risk of conflict due to reduced opportunity costs of fighting.

## 3 Data

The Locust Watch database (FAO 2022) includes data from 1985 to the present on observations of desert locust swarms, as well as smaller concentrations of locusts. These data include latitude, longitude, and date of observations. I consider only data on locust swarms, high density groups of gregarious locusts that can move in a coherent manner, and do not consider observations of locusts at lower density as these typically pose less of a threat.

Data on conflict events come from the Armed Conflict Location & Event Data Project (ACLED) database (Raleigh et al. 2010). The database records the location, date, and nature of conflict events globally starting from 1997. The analysis focuses on events categorized

by ACLED as "violent conflict,", which includes battles, explosions, and violence against civilians. I test robustness to analyzing protest and riot events and to using data on violent conflicts from the Uppsala Conflict Data Program (UCDP) (Sundberg and Melander 2013), which uses a more restrictive definition of conflict.<sup>3</sup>

I collapse the data to raster grid with annual observations for cells with a 0.25° resolution (15 arcminutes, approximately 28km<sup>2</sup>). In each cell and year I measure whether any locust swarm/conflict event was observed. To account for possible spatial spillovers, I also measure whether any swarms are observed in bands at different distances outside of the cell.I categorize swarms as arriving during particular stages of the crop production cycle by matching the specific date a swarm is observed to country-level crop calendars for staple grains and main cash crops from The United States Department of Agriculture (USDA) (2022).<sup>4</sup> I define four main seasons: planting, growing, harvesting, and the off season between harvesting and planting. Figure A2 shows the share of sample cells at different stages of agricultural cycle by month and the counts of locust swarms observed by season and region.

Given the role of weather in desert locust biology and its importance in determining agricultural production, all analyses control for local weather to isolate the impact of the arrival of a locust swarm. I measure annual precipitation (in mm) and maximum temperature (in °C) through 2018 using high-resolution monthly data from WorldClim.<sup>5</sup> I also incorporate raster population data for every 5 years from CIESIN 2018, linearly interpolating within cells between years where the population is estimated, and raster data on land cover in 2000 from CIESIN, giving the share of land cover that is cropland and pasture (Ramankutty et al. 2010).

Although desert locust swarms are observed as far east as India, I restrict my analysis to countries in Africa and the Arabian Peninsula with at least 10 locust swarm observations from 1985-2018. These countries include all of North Africa, most of the Arabian Peninsula,

 $<sup>^{3}</sup>$ UCDP records conflicts worldwide since 1989 involving at least one "organized actor" and resulting in at least 25 battle-related deaths in a calendar year. ACLED has no organized actor or minimum death threshold requirements.

<sup>&</sup>lt;sup>4</sup>Figure A1 shows example crop calendars from Libya and Mali. In countries with different agricultural cycles by crop, I identify the crop activity associated with the most commonly grown crops each month.

<sup>&</sup>lt;sup>5</sup>CRU-TS 4.03 (Harris et al. 2014) downscaled with WorldClim 2.1 (Fick and Hijmans 2017).

and countries along the Sahel. I drop unpopulated (largely desert and water) cells from the analysis. Since ACLED records conflicts beginning in 1997 and the weather data are available until 2018, I retain only data from 1997 to 2018.

The resulting analysis sample covers 22 years across 24,459 cells, for a total of 538,086 observations. Among these are 2,634 cell-years with a locust swarm event and 10,265 with a violent conflict event. Ten percent of cells in the sample experienced at least one locust swarm, and 56% were within 100km of at least one locust swarm event. Fourteen percent of cells experienced at least one violent conflict event. About half the cells (53%) in the sample include some agricultural land: 52% have pasture land while 28% have crop land. Across all cells, mean pasture area is 19% of the cell and mean crop area is 5% of the cell. These variables are displayed in Figure 2, and summary stats are included in Table A1. I conduct my main analyses using the full analysis sample, and test robustness and heterogeneity using subsamples based on these characteristics.

#### [Figure 2 here]

### 4 Empirical approach

I estimate the causal impacts of locust swarms on conflict in the short term using a linear probability model estimated via OLS, which takes the form:

$$Conflict_{cit} = \alpha + \beta Swarms_{cit} + \delta X_{ct} + \gamma_{it} + \mu_c + \epsilon_{cit}$$
(1)

where c indexes cells, i indexes countries, and t indexes years. Conflict is a dummy variable for observing any conflict event and Swarms is a dummy variable for observing any locust swarm.  $\gamma_{it}$  are country-year fixed effects, and  $\mu_c$  are cell fixed effects.  $X_{ct}$  is a vector of controls at the cell level. My preferred specification includes as controls an indicator for any locust swarms in the area outside the cell within 100km from the cell centroid, total annual rainfall (in mm), the maximum annual temperature (in °C), and 1 year lags of locust swarms, rainfall, and max temperature. Standard errors (SEs) are clustered at the country level to allow for correlation in the errors within countries.<sup>6</sup>

This fixed effects model is similar to Fjelde (2015), Harari and La Ferrara (2018), McGuirk and Nunn (2021), and Ubilava, Hastings, and Atalay (2022), and others in the use of a grid cell panel data to analyze the impact of weather on conflict in Africa and to Burke, Hsiang, and Miguel (2015) in structure and the use of lagged weather variables. The countryyear fixed effects flexibly control for factors varying over time at the country level that might affect conflict and the impact of locust swarms, such as the policy environment and national economic and social conditions. The cell fixed effects control for time invariant cell characteristics, such as geography and typical wind patterns. Effects of locusts are therefore identified from variation in swarm presence within cells over time controlling for time-varying national conditions.

Controlling for swarms in the previous year and in the area outside the cell accounts for potential temporal and spatial spillovers. The rainfall and temperature controls and lags allow me to isolate the impact of the locust shock from concurrent environmental factors that may affect agricultural productio and the risk of conflict. Desert locust outbreaks follow periods of heavy rainfall and vegetation growth in breeding areas. Given spatial correlation in weather, this would tend to increase agricultural production in affected areas if not for the destruction of locust swarms. Indeed, while swarms cause major localized agricultural losses, at the national level production may increase in outbreak years (Krall and Herok 1997).

Conditional on swarm formation in breeding areas, variation in wind direction and typical locust flight duration create quasi-random variation in areas where swarms land. After including controls for weather and fixed effects, we can therefore consider swarm shocks to be exogenous to local conditions which might affect the risk of conflict and interpret the coefficient on *Swarms* as a causal impact.

<sup>&</sup>lt;sup>6</sup>This is likely more restrictive than necessary and will lead to a conservative interpretation of the results. Results are similar when using Conley (1999) Heteroskedasticity and Autocorrelation-Consistent (HAC) SEs allowing for both spatial and serial correlation following Hsiang (2010)'s approach.

I test robustness of the results to different controls and fixed effects, to different outcome definitions, to different restrictions of the analysis sample, and to different clustering of standard errors. Results of robustness tests are included in Appendix D. To test for heterogeneity in the impacts of swarms, I estimate Equation 1 fully interacting the right-hand side variables with another variable of interest. I test for spatial spillovers by considering impacts of swarms in bands at a particular radius from the cell, and by estimating impacts at different levels of analysis, collapsing the data across cells. To test whether effects vary by swarm timing, I estimate Equation 1 separating out *Swarms* into a series of dummy variables indicating the presence of locust swarms during particular periods of the crop calendar.

Finally, to test whether impacts of locust swarms persist beyond the short term I analyze long-term impacts of the 2003-2005 locust upsurge, the last major locust outbreak prior to the most recent upsurge in 2019-2021 and the only major upsurge in the sample period (1997-2018). This upsurge accounts for 59.5% of swarm observations in the sample. I estimate a two-way fixed effects difference-in-differences regression

$$Conflict_{cit} = \alpha + \beta Swarms_{cit} + \xi Upsurge_{ci} \times Post_t + \delta X_{ct} + \gamma_{it} + \mu_c + \epsilon_{cit}$$
(2)

where *Upsurge* is an indicator for being in a cell with any locust swarm between 2003-2005 and *Post* is an indicator for being in a year after 2005. The fixed effects absorb the individual *Upsurge* and *Post* terms. This is a 'canonical' difference-in-differences' analysis with the upsurge 'treatment' occurring in the same period for all treated units and a comparison group that never receives this treatment. I also conduct an event study analysis of the upsurge replacing *Post* with individual year dummies.

Identification for the analysis of long-term impacts relies on the assumption of parallel trends between areas that did and did not experience locust swarms during the 2003-2005 upsurge. This assumption is supported by the quasi-random variation in where locusts land due to wind speed, direction, and flight duration. I also test for parallel pre-trends using the event study specification, and test the robustness of the results to different constraints on the areas included in the comparison sample.

## 5 Results

#### 5.1 Short-term impacts

Table 1 presents estimates of Equation 1 for different subsamples, separately analyzing impacts on violent conflict events. The point estimates for contemporaneous and lagged weather are almost uniformly positive across subsamples: deviations from mean annual temperature and rainfall within cells are associated with a higher probability of conflict, consistent with the literature on climate and conflict. The magnitudes of the effects of rainfall and temperature fall in the upper middle of the range of estimates reported in Burke, Hsiang, and Miguel (2015)'s meta-analysis of the impacts of weather deviations on intergroup conflict.

Effects of rainfall and temperature are not always statistically significant when clustering SEs at the country level, but are significant when using Conley SEs allowing for more flexible spatial correlation. I find that SEs clustered at the country level are uniformly larger than SEs clustered at the country-year or cell level, and than Conley (1999) SEs allowing for spatial correlation within a radius of 500km (Figure A7). This is expected given that clustering at the country level implies a quite large level of spatial and serial correlation. SEs clustered at the country-year level are only slightly smaller on average than the SEs clustered at the country level, indicating spatial correlation in the errors is relatively more important than serial correlation in these analyses. I report only the country-clustered SEs in the main results as these are more conservative, though this approach might understate the significance of certain relationships.

#### [Table 1 here]

In contrast to rainfall and temperature, locust swarms significantly decrease the proba-

bility of conflict in the same year. In cells where a locust swarm is observed in a given year, the probability of observing any violent conflict event in that cell in that year falls by 1.5 percentage points holding all else constant in the full sample. This represents a reduction of 76% relative to the mean probability of observing violent conflicts in cells with no locust swarms. This result is robust to several restrictions of the set of analysis cells, shown in columns (2)-(6), and in particular to restricting the set of control cells to those that ever experienced a locust swarm during the sample period (column 5) and those that ever experienced violent conflict (column 6). The results from these subsamples indicate that locust swarms more conservatively decrease the risk of conflict by around 20% relative to cell-years with no swarms. While smaller than the estimate across all sample cells, which may be an upper bound, this remains a large and significant effect.

Experiencing a locust swarm the previous year also reduces the risk of violent conflict, by 0.9 percentage points (45%) in the full sample. This result loses statistical significance in the most restrictive samples, but the similar sign for both current and lagged swarms suggest a similar mechanism may be driving the impacts on the risk of conflict.<sup>7</sup>

The negative impact of locust swarms on conflict risk in the same year is robust to a variety of different specifications. Point estimates are consistently negative in specifications varying the set of control variables and fixed effects, but are only statistically significant when including weather controls (Table A3). Results are robust to varying the size of cells up to the level of 2 degree cells (Table 2), which I return to in the discussion of conflict spillovers. The proportional impact of locust swarms is similar in the samples of years before and after 2010, when conflict frequency began to markedly increase in the study area, though the post-2010 estimate is noisy as fewer swarms are observed in this period. Results are robust to dropping different regions of the study area from the analysis, indicating results are not driven by any one region (Table A4). Results are similar when considering alternative measures

<sup>&</sup>lt;sup>7</sup>Both Crost et al. (2018) and Harari and La Ferrara (2018) find that a negative agricultural shock increases the risk of conflict in the current and following year, so similarly find persistent effects in the short-term but with the opposite sign.

of conflict, including ACLED protest/riot events, the more restrictive UCDP definition of violent conflict events, whether a state or government actor is involved in the conflict, and the intensive margin using counts of fatalities from conflict events in a year (Table A5).<sup>8</sup> Finally, I find no significant difference in the effect of experiencing a single swarm in a given year as opposed to multiple swarms on the risk of conflict across a variety of specifications, validating the focus of the analysis on the extensive margin of locust presence rather than the intensive margin (Figure A8).

A concern might be that violent conflict reduces the probability that locust swarms are observed, but the results are robust to controlling for this possibility, as shown in Table A2. I find no effect of violent conflict in the first half of a year on the probability of observing a locust swarm in the second half of the year (column 2), though conflict the previous year is associated with a lower probability of observing a swarm the current year (column 1). The effect of a locust swarm remains significant and large when controlling for lagged conflict, causing a 60% decrease in the risk of conflict in the full sample of cells (column 4) and is similar when restricting the sample to observations with no conflict the prior year (column 5). Further, a locust swarm in the first half of the year decreases the risk of violent conflict in the second half of the year by 42% (column 6.Reverse causality therefore does not apper to drive the negative effect of locust swarms on conflict.

#### 5.2 Spatial spillovers

One possible explanation for the decrease in conflict risk in cells affected by locust swarms is spatial conflict spillovers. The returns to conflict will be lower in areas affected by swarms, making nearby unaffected areas relatively more attractive to groups aiming to capture resources. This could be particularly true since years with locust swarm outbreaks are typically more productive agriculturally, as the rainfall shocks in locust breeding areas leading to

<sup>&</sup>lt;sup>8</sup>The impact of locust swarms on conflict when using UCDP data is only significant when considering impacts on any conflict event. Estimated impacts on specific types of conflict are negative but imprecise given the limited number of UCDP conflict events in the sample.

swarm formation are often correlated with positive rainfall shocks in agricultural areas. If conflict risk increases in areas surrounding locust-affected cells, the net effect of locusts on conflict at a broader spatial level might be null or positive.

As an example of this, McGuirk and Nunn (2021) find that droughts in pastoral areas of Africa lead to conflict in neighboring agricultural areas as pastoral groups are displaced in search of grazing opportunities. These spillovers lead to a negative estimated relationship between precipitation shocks and conflict in the sample overall, since the conflict takes place outside affected areas.Showler (2019) report instances of similar resource-based conflicts between farmers and pastoralists as a consequence of population movements caused by the 2003-2005 locust upsurge in West Africa, indicating potential for such conflict spillovers.

The main regression specification includes as a control an indicator for any locust swarm observed within 100km outside the cell (approximately 28km<sup>2</sup>). Table 1 shows that locust swarms in the 100km outside a cell do not have any effect on the risk of conflict within the cell: the point estimate is a fairly precise 0. Figure 3 shows that this result is not sensitive to the choice of distances outside the cell to consider. Coefficients are close to zero and generally non-significant for swarms in bands at increasing distances from a cell in both the current and previous year. An exception is that locust swarms within 50km outside a cell and 100-150km outside a cell are marginally significantly associated with 0.4 and 0.2 percentage point *decreases* in the likelihood of violent conflict, respectively. If anything, this suggets that spillovers of swarm presence further suppress the risk of conflict in nearby areas, rather than displacing conflict to those areas.<sup>9</sup>

#### [Figure 3 here]

Another approach to testing whether spillovers may affect the results to consider whether estimates vary with the granularity of the analysis, as in McGuirk and Nunn (2021). Table 2

 $<sup>^{9}</sup>$ A possible reason for the negative effect of swarms immediately outside the cell is that those swarms may also have caused damages inside the cell. Swarms are assigned to cells based on the coordinates at which they are reported, but locust swarms can cover from under 1km<sup>2</sup> to several hundred at their largest extent, so some may cross into adjacent cells.

presents results from estimating the main specification at different scales. I collapse the data to higher levels of aggregation by taking the maximum of swarm and conflict event dummies and means of weather variables across  $0.25^{\circ}$  cells within the aggregated area. For example, in Column (2) both the violent conflict and swarm event variables measure whether such an event was recorded in any of the four  $0.25^{\circ}$  cells within a  $0.5^{\circ}$  cell.

#### [Table 2 here]

Estimated impacts of locust swarms on conflict are negative and statistically significant when aggregating cells up to  $1^{\circ}$  (around  $110 \text{km}^2$ ), remain negative but no longer significant for  $2^{\circ}$  cells, and are positive and non-significant at the 5° cell or country level. Absolute effect magnitudes are increasing in the level of analysis up to 1° cells, though impacts relative to the mean conflict risk in areas with no swarms are decreasing as the likelihood that areas experience any conflict increases with the size of the area. For example, any locust swarm reported in a 1° cell decreases the probability of experiencing violent conflict in that year by 3.1 percentage points, or 24% relative to the mean in areas with no swarms.

These results are consistent with negative effects concentrated within cells and no significant spillovers in areas up to 250km away; conflict is not simply being displaced from the area affected by locusts to another nearby area. Decreases in relative impact magnitude at higher aggregations likely result from reduced treatment intensity, as the share of total area affected by locusts within treated areas falls at higher levels of aggregation. Figure 3 suggests it is not driven by positive conflict spillovers which offset the local decreases in conflict risk.

Positive non-significant effects of locusts on conflict at the 5° and country level likely reflect further reductions in locust treatment intensity as well as lower variation in the probability of conflict at these levels. When taking the mean instead of the maximum for conflict and swarm events across  $0.25^{\circ}$  cells within the aggregated areas to preserve treatment intensity, point estimates are negative and non-significant at the 5° cell and country level, and the negative effect at the 2° level becomes statistically significant (Table A6). The signs for the estimated impacts of temperature deviations on conflict risk also change at higher levels of aggregation, from positive to negative, suggesting aggregating variables across such large geographic areas loses too much of the spatial variation and makes it challenging to estimate causal relationships.

#### 5.3 Agricultural destruction and seasonality

The impacts of locust swarms on conflict should operate primarily through first order effects on agricultural output. Locusts do not cause direct damages outside of consuming vegetation, though secondary impact channels could include psychological impacts or potential negative externalities from efforts to prevent crop destruction such as poisoning from pesticides or exposure to smoke from fires aiming to deter locusts.

Table 3 shows that impacts of locust swarms are indeed concentrated in agricultural areas, based on measures of land cover in 2000 from Ramankutty et al. (2010).<sup>10</sup> Point estimates for cells with no crop land or pasture land are negative but not significant.

#### [Table 3 here]

The impacts are largest in areas with crop land which are relatively more conflict-prone. Swarms decrease the risk of conflict by 2.4 percentage points more in cells with crop land compared to cells without, a 4.8 times larger effect, leading to a 62% decrease in conflict risk. The impact of swarms is 2 times larger in cells with pasture land than in cells with none, decreasing the risk of conflict by 53%. The effect of swarms the previous year remains negative and is larger in magnitude for agricultural areas, but not significantly so. There is no significant impact on conflict of swarms in the area surrounding a cell either in the current or previous year, regardless of whether the cell has agricultural land.

These results indicate that impacts of swarms on agricultural land and cropland in particular are the primary driver of the overall negative effect of swarms on conflict. I next

<sup>&</sup>lt;sup>10</sup>Southward expansion of the Sahara desert, anti-desertification efforts, deforestation, changing seasonal distribution of precipitation, and expansion of farming in traditional pastureland have all contributed to changing land cover of the study period (Davis 2022; Liu and Xue 2020; Rahimi et al. 2021). If agriculture expanded to some areas with no pasture or crop land after 2000 (relatively early in our 1997-2018 sample period), this might reduce the estimated difference in impact by cell land cover.

estimate differences by swarm timing relative to the crop cycle, building on previous research on crop seasonality and conflict. Crost et al. (2018) find that the impact of increased rainfall on conflict in the Philippines depends on the season because of different effects on agriculture: a positive rainfall shock decreases conflict risk by increasing the opportunity cost of fighting. Guardado and Pennings (2021) show that the onset of harvest in Afghanistan, Iraq, and Pakistan similarly reduces conflict risk by changing the opportunity cost. If the negative impact of swarms on the risk of conflict is due to the returns to conflict falling by more than the opportunity costs, we should expect to find larger negative impacts of swarms arriving in parts of the year when crop destruction will be largest: the growing and early harvest months. Off-season swarms should have limited effects.

Table 4 presents the estimated impacts of swarms arriving in different seasons for subsets of cells by land cover. Cells with any crop land (nearly all of which also include pasture land) account for 28.3% of the sample and 36.2% of swarm observations. The count of swarms observed across different points in the agricultural cycle is similar, though somewhat higher in the growing and harvest seasons than in the off or planting seasons (Figure A2). Seasonality may also be relevant in cells with pasture land; agricultural cells together account for a further 75.1% of swarm events.

As in Table 1, prior year swarms consistently reduce the risk of conflict while effects of swarms in the 100km area outside the cell are not statistically significant. Consistent with a negative overall impact of swarms on the risk of conflict, point estimates for the impacts of swarms arriving in different seasons are negative, with the exception of swarms arriving during the harvest period where point estimates are positive but close to 0.

#### [Table 4 here]

Swarms arriving the off-season between harvest and planting of major crops significantly decrease the risk of conflict in cells with crop area, by 83%. This drives a large average effect across all cells. The difference in impacts in crop cells and agricultural cells in general indicates off-season swarms have a negligible impact in pastoral areas. This is surprising:

there should be no or very limited crop destruction during the off season, while destruction of existing vegetation might have been expected to adversely affect pastoralists.

Although the estimated impact magnitude in crop cells is largest for off-season swarms in crop cells, in all cells on average the largest magnitude is for planting season swarms: a 1.8 percentage point decrease in conflict risk. The magnitude is similar to those for crop cells and all agricultural cells, where estimates are close to marginally significant (p = 0.102and p = 0.124, respectively). Similar estimated magnitudes for swarm impacts on conflict risk across types of land cover suggest this effect may not relate to agricultural damages, particularly since damage is likely to be limited during the parts of the planting season before seeds sprout. Growing season swarms do not significantly affect the risk of conflict, and the magnitudes of the point estimates are much smaller than for off-season or planting season swarms across all types of land cover.

The pattern of results is similar when analyzing impacts at the level of 0.5° cells rather than 0.25° (Figure A9), though at this level the impacts of planting season swarms on conflict are slightly smaller and no longer statistically significant while the impacts of growing season swarms are larger and significant in all samples. Estimated effects by season are also similar across regions, despite differences in crop calendars (Figure A10). This indicates that impacts by season are due to real differences in locust effects along the crop cycle rather than potentially mechanical differences by month of year.

Table 4 thus illustrates significant heterogeneity in swarm impacts by timing relative to the agricultural calendar, but in unexpected ways that do not align with predictions based on changes in returns to agriculture and to fighting.<sup>11</sup> Controlling for whether swarms arrive in other seasons of the year,<sup>12</sup> swarms arriving in the off season should have a limited effect

<sup>&</sup>lt;sup>11</sup>Variation in the effects of prior year swarms by season also differ from the pattern that would be expected if lagged swarm impacts on current conflict risk were due to decreased returns to fighting (Figure A10). Negative overall effects are driven by swarms that arrived during the previous year's planting season. Point estimates for impacts of growing season swarms the prior year, when crop destruction should be most severe, are positive and close to significant at the 10% level. Point estimates for prior year harvest season swarms are close to 0.

<sup>&</sup>lt;sup>12</sup>This is an important consideration, as there is some serial correlation in risk from locust swarms, particularly during major outbreaks or upsurges. Table A2 shows that locations with a swarm the prior year

on crop production, yet estimated impacts of swarms in cells with any crop land are largest for off-season swarms. This could reflect destruction of tree crops and other perennials, but annual crops (such as staple grains) take up the large majority of crop land in the sample countries overall so swarm damages to perennial crops should be small relative to damages to annual crops. Most cells with crop land also include some pasture land, so off-season impacts may also be due to loss of fodder for livestock—potentially very important as the off season in most countries is drier making sources of animal feed more scarce. But the coefficient for off-season swarms is smaller in agricultural cells overall than in crop cells in particular, suggesting that effects on pasture do not drive off-season swarm impacts.

Agricultural destruction should be largest for growing season swarms, as crops have sprouted but not yet been harvested. This implies that impact magnitudes should be largest for swarms arriving in this season, if agricultural destruction is the main mechanism. Results using  $0.5^{\circ}$  cells are more consistent with this, as the negative magnitudes are larger for growing season swarms than for planting or harvest season swarms (Figure A9). But with the main analysis sample of  $0.25^{\circ}$  cells, the estimated impacts are larger for planting season swarms than for growing season swarms.

Depending on whether harvest is completed by the time locusts arrive, swarms in the harvest season could either lead to very large agricultural destruction or have no effect. Assuming that harvest season swarms do reduce agricultural output on average, if this is the mechanism driving impacts of swarms on conflict we should expect a negative effect as seen for swarms in other seasons, but the point estimates for harvest season swarms are *positive*—though not distinguishable from 0.

These results highlight the importance of seasonality in the impacts of locust swarms on conflict risk, but a simple model of conflict based on changes in the returns and opportunity costs of fighting struggles to explain the patterns of impacts. This indicates that the mechanisms driving conflict responses to agricultural shocks are more complex than is presented

are less likely to have a locust swarm the current year, but locations with a swarm in the first half of the year are more likely to have a locust swarm in the second half of the year.

in much of the economics literature.

#### 5.4 Long-term impacts

The analyses thus far have focused on the short-term: swarms in both the prior year and the same year reduce the likelihood of violent conflict events. But Table 1 shows that the probability of any violent conflict in years with no swarm is greater among cells that ever had a swarm than in cells that did not. Could this difference be due to positive long-term impacts of swarms on conflict risk?

I test long-term effects of locust swarms by considering impacts of the major locust upsurge in 2003-2005, the main outbreak in the sample period which affected 6.6% of cells. Table 5 presents the results from estimating Equation 2. Controlling for the 2003-2005 locust upsurge, locust swarms in the current and previous year still significantly reduce the risk of any violent conflict event, though the estimated effects of a prior year swarm are now larger than those of a swarm in the current year. In contrast, the 2003-2005 locust upsurge *increases* the risk of conflict in the following years. Cells where swarms were reported during this upsurge are 1.6 percentage points (62%) more likely to experience violent conflict in in a given after year 2005 relative to cells that were not affected by this upsurge. The results are nearly identical in the full sample and in the subsamples of cells within 250km and 100km from any swarm during the upsurge, indicating the effect is not driven by comparing upsurge-affected areas to dissimilar areas.

#### [Table 5 here]

This result suggests that locust swarms decrease the opportunity cost of fighting in the long term.<sup>13</sup> Adoption of agricultural insurance is low in the sample countries, and local risk sharing networks offer less support for a broad common shock such as a locust swarm. Recovery from locust shocks in this setting may therefore be limited. Households use a

<sup>&</sup>lt;sup>13</sup>It is possible but highly unlikely that they increase the long-term returns to conflict.

variety of measures to cope with short-term food security and livelihood effects of locust outbreaks. Selling assets, sending household members away, and consuming seed stocks are commonly reported coping strategies that would adversely impact agricultural production in following years, decreasing the opportunity cost of fighting.

One puzzle is that this mechanism would predict increases in the risk of conflict to be greatest the year after a locust swarm arrives, as this is when household coping mechanisms would be expected to most adversely affect agricultural production. But the results consistently show negative effects of swarms in the previous year on the probability of conflict of a similar magnitude as effects of swarms in the current year. This indicates that either the effect of the upsurge on opportunity costs of fighting is delayed in some way, or that there are other mechanisms involved, or both.

Figure 4 shows the results of an event study analysis of the 2003-2005 upsurge. There are no significant differences in the risk of conflict between areas affected by locust swarms during this upsurge and areas that were not in the years preceding the upsurge (p = 0.755), and point estimates are close to 0. This supports the assumption of parallel trends between these areas if not for the upsurge. The analysis controls for whether any swarms were observed in the current and prior year. Consequently, there are no significant impacts of the upsurge in 2004 and 2005, the main years of the upsurge.<sup>14</sup>

#### [Figure 4 here]

Estimated impacts of the 2003-2005 upsurge on conflict become larger in magnitude over time, increasing in a nearly linear fashion from 2005 on except for a dip in 2018 at the end of the sample period. As would be expected, the standard errors increase over time with greater separation from the event, but the estimates are significant at the 90% level or greater for all years after 2008, including at the 95% confidence level for 2010, 2012, 2014, 2016, 2017, and 2018. Being affected by locust swarms during the 2003-2005 upsurge increases the risk

 $<sup>^{14}</sup>$ The coefficient for 2004, for example, would therefore be interpreted as the impact on conflict of having been affected by a swarm in 2005.

of conflict in each year from 2011-2018 by between 2.2 and 4.4 percentage points, controlling for current and prior year locust swarms, weather, and cell and country-by-year fixed effects.

Though the magnitude of the impact of the upsurge increases over time, the size of the effect relative to the probability of violent conflict in unaffected areas generally decreases because that probability has been increasing over time, due to a variety of factors (Figure A3). The 2.8 percentage point increase in conflict risk in 2011 is a 171% increase relative to the risk in non-affected areas, while the 4.4 percentage point increase in 2017 is a 75% relative increase and the 3.3 percentage point increase in 2018 is a 59% relative increase.

The results are nearly unchanged when considering only the cells within 250km of a swarm during the upsurge and when restricting the sample to only cells with any agricultural land in 2000 (Figure A11). The pattern of increasing conflict risk over time also holds when using different definitions of conflict (Figure A12). There is a clear long-term increase in conflict risk in areas affected by the upsurge, particularly after 2010.

## 6 Conclusion

While desert locusts can have devastating consequences for local agriculture, this analysis shows that the arrival of a locust swarm does not increase the risk conflict. Instead, locust swarms *decrease* the likelihood of experiencing any violent conflict event in a given year by around 20% after controlling for the effects of rainfall, temperature, time-invariant local characteristics, and country-by-year fixed effects. Impacts of swarms on conflict are largely local with no significant spillovers into surrounding areas.

Swarms decrease the risk of conflict much more in agricultural areas with effects on crop land particularly large. A limitation of this analysis is the use of land cover data from 2000. Data on land cover over time would help to clarify the importance of crop versus pasture damages as mechanisms for the impacts of locust swarms. Incorporating measures of agricultural destruction following swarm events, potentially using satellite data on changes in vegetation, could be used to estimate direct impacts of agricultural damage on conflict.

Differences in the impact of swarms on conflict by their timing relative to the crop growing calendar indicate that effects are not driven solely by changes in agricultural productivity. Changes in opportunity costs related to non-agricultural activities by time of year could explain the patterns in effects of locusts on conflict risk. Farmers may see off and planting season locust swarms as indicating an increased risk of swarms later in the agricultural cycle and be better situated to switch into non-farm employment, whether that be a household enterprise, local wage work, or migrating in search of work. The availability of non-agricultural work sets a lower bound on how much the opportunity cost of fighting can fall following an agricultural shock, and may explain the negative effects of locusts on conflict risk as agricultural destruction also decreases the returns to conflict.

One potential implication of this result is that households would prefer to respond to an agricultural shock by engaging in a productive activity to earn their livelihood rather than to engage in conflict. This would imply that policies to increase the diversity and resilience of livelihood strategies in the sample countries could decrease the risk of violent conflict following an adverse agricultural shock, as has been shown by some recent studies (Fetzer 2020; Garg, McCord, and Montfort 2020).

The contemporaneous reduction in conflict is contrary to what we might expect from a food insecurity shock, but major locust outbreaks are the object of significant international attention and aid which may attenuate food insecurity effects in the short term. Individuals in locust-affected areas may also attempt to avoid violent conflict in the short term in order not to deter potential aid from arriving. Geographically disaggregated data on locust relief efforts (or more general aid flows) could be useful to explore whether these play a role in reducing risk of fighting by increasing its opportunity cost in both the short and long term. Data on food insecurity could help identify whether impacts on conflict differ when locust swarms have more adverse effects on food security. Migration is also a common response to locust shocks: over 8 million people were displaced across East Africa as a result of the 2019-2021 locust outbreak (The World Bank 2020). Data on population movements could help test this as a mechanism for negative local impacts of locust swarms.

Although short-term impacts of swarms on the risk of violent conflict are negative, the long-term impact is positive indicating that locust damages have permanent effects despite the transient nature of swarms. Areas affected by the 2003-2005 locust upsurge are 62% more likely than unaffected areas to experience violent conflict in a given year after 2005. Many factors have contributed to a general increase in violent conflict in the sample countries in this period, including several civil wars, insurgencies, and the spread of terrorist organizations. Locust swarms appear to make communities particularly vulnerable to engaging in these conflicts, potentially due to a permanent decrease in agricultural productivity.

Long-term negative impacts of locust swarms on children's education (De Vreyer, Guilbert, and Mesple-Somps 2015) and health (Conte, Tapsoba, and Piemontese 2021) could explain part of the positive long-term effect on conflict, but analyzing long-term changes in agricultural productivity, labor supply, household wealth, migration, and psychological factors such as aspirations and beliefs would help elucidate the key mechanisms. Future work could also incorporate data on other types of agricultural shocks and explicitly compare impacts on conflict with those of locust swarms to highlight what makes locust shocks different, as well as test whether a past locust shock makes households more vulnerable to another agricultural shock. In general, these results highlight how failing to support communities affected by disasters to fully recover can create conditions for future conflict.

Beyond contributing to our understanding of the relationship between agricultural productivity shocks and conflict risk, the findings are also relevant for considering multilateral policy around desert locusts. Locusts do not respect country boundaries and require international coordinate for adequate monitoring and control. This paper highlights another important consequence of desert locust outbreaks which should be considered in weighting the costs and benefits of locust monitoring and control operations.

## References

- ASU. 2020. <u>The Global Locust Initiative</u>. Arizona State University, https://sustainability.asu.edu/global-locust-initiative/.
- Baker, Richard B, John Blanchette, and Katherine Eriksson. 2020. "Long-run impacts of agricultural shocks on educational attainment: Evidence from the boll weevil". The Journal of Economic History 80 (1): 136–174.

Banerjee, Abhijit, et al. 2010. "Long-run health impacts of income shocks: Wine and phylloxera in nineteenth-century France". <u>The Review of Economics and Statistics</u> 92 (4): 714–728.

- Blattman, Christopher, and Edward Miguel. 2010. "Civil war". Journal of Economic literature 48 (1): 3–57.
- Botzen, WJ Wouter, Olivier Deschenes, and Mark Sanders. 2019. "The economic impacts of natural disasters: A review of models and empirical studies". <u>Review of Environmental Economics and Policy</u>.

Brader, L, et al. 2006. "Towards a more effective response to desert locusts and their impacts on food security, livelihoods and poverty".Multilateral evaluation of the 2003–05 Desert locust campaign. Food and Agriculture Organisation, R

- Bradshaw, Corey JA, et al. 2016. "Massive yet grossly underestimated global costs of invasive insects". Nature communications 7 (1): 1–8.
- Burke, Marshall, Solomon M Hsiang, and Edward Miguel. 2015. "Climate and conflict". Annual Review of Economics.
- Cavallo, Eduardo, et al. 2013. "Catastrophic natural disasters and economic growth". Review of Economics and Statistics 95 (5): 1549–1561.
- Chassang, Sylvain, Gerard Padró i Miquel, et al. 2009. "Economic shocks and civil war". Quarterly Journal of Political Science 4 (3): 211–28.
- Chemin, Matthieu, Joost De Laat, and Johannes Haushofer. 2013. "Negative rainfall shocks increase levels of the stress hormone cortisol among poor farmers in Kenya". Available at SSRN 2294171.

#### CIESIN. 2018.

Gridded Population of the World, Version 4 (GPWv4): Population Count, Revision 11. Center for International Earth Science Information Network - CIESIN - Columbia University. Palisades, NY: NASA Socioeconomic Data and Applications Center (SEDAC). https://doi.org/10.7927/H4JW8BX5.

- Collier, Paul, and Anke Hoeffler. 2004. "Greed and grievance in civil war". Oxford economic papers 56 (4): 563–595.
- . 1998. "On economic causes of civil war". Oxford economic papers 50 (4): 563–573.
- Conley, Timothy G. 1999. "GMM estimation with cross sectional dependence". Journal of econometrics 92 (1): 1–45.

- Conte, Bruno, Augustin Tapsoba, and Lavinia Piemontese. 2021. "The Power of Markets: Impact of Desert Locust Invasions on Child Health".
- Cressman, Keith, and Cyril Ferrand. 2021. <u>Signs of hope in East Africa, as control campaign tames locust upsurge</u>. Food and <u>Agricultural Organization, https://www.fao.org/news/story/en/item/1393635/icode/</u>.
- Cressman, Keith, and Robert Stefanski. 2016. <u>Weather and Desert Locusts</u>. Food and Agricultural Organization.
- Cressman, Keith, Alice Van der Elstraeten, and Clare Pedrick. 2016. eLocust3: An innovative tool for crop pest control. Food and Agricultural Organization.
- Crost, Benjamin, et al. 2018. "Climate change, agricultural production and civil conflict: Evidence from the Philippines". Journal of Environmental Economics and Management 88:379–395.
- Dal Bó, Ernesto, and Pedro Dal Bó. 2011. "Workers, warriors, and criminals: social conflict in general equilibrium". Journal of the European Economic Association 9 (4): 646–677.
- Dal Bó, Ernesto, and Robert Powell. 2009. "A model of spoils politics". American Journal of Political Science 53 (1): 207–222.
- Davis, Marion. 2022. <u>Pastoralism, farming and a changing climate in the Sahel region</u>. Stockholm Environment Institute (SEI). https://www.sei.org/featured/pastoralism-farming-climate-in-sahel/.
- De Vreyer, Philippe, Nathalie Guilbert, and Sandrine Mesple-Somps. 2015. "Impact of natural disasters on education outcomes: evidence from the 1987–89 locust plague in Mali". Journal of African Economies 24 (1): 57–100.
- Dell, Melissa, Benjamin F Jones, and Benjamin A Olken. 2012. "Temperature shocks and economic growth: Evidence from the last half century". American Economic Journal: Macroeconomics 4 (3): 66–95.
- Dobson, Hans M. 2001. <u>Desert locust guidelines: control</u>. Food and Agriculture Organization.
- Dube, Oeindrila, and Juan F Vargas. 2013. "Commodity price shocks and civil conflict: Evidence from Colombia". The review of economic studies 80 (4): 1384–1421.
- Fearon, James D. 1995. "Rationalist explanations for war". International organization 49 (3): 379–414.
- Fetzer, Thiemo. 2020. "Can workfare programs moderate conflict? Evidence from India". Journal of the European Economic Association 18 (6): 3337–3375.
- Fick, Stephen E, and Robert J Hijmans. 2017. "WorldClim 2: new 1-km spatial resolution climate surfaces for global land areas". <u>International journal of climatology</u> 37 (12): 4302–4315.
- Fjelde, Hanne. 2015. "Farming or fighting? Agricultural price shocks and civil war in Africa". World Development 67:525–534.

- Food and Agriculture Organization of the United Nations (FAO). 2022a. Desert Locust upsurge may be declining but remaining swarms require vigilance in East Africa and Ye https://www.fao.org/newsroom/detail/desert-locust-upsurge-may-bedeclining-but-remaining-swarms-require-vigilance-in-east-africa-andyemen/en.
- 2022b. Locust Watch: Desert Locust.
   Food%20and%20Agriculture%200rganization%20of%20the%20United%20Nations%
   20(FAO),%20http://www.fao.org/ag/locusts/en/info/info/index.html.
- French, Eric, and Christopher Taber. 2011. "Identification of models of the labor market". In Handbook of labor economics, 4:537–617. Elsevier.
- Garg, Teevrat, Gordon C McCord, and Aleister Montfort. 2020. Can Social Protection Reduce Environmental Damages? IZA DP No. 13247.
- Gignoux, Jérémie, and Marta Menéndez. 2016. "Benefit in the wake of disaster: Long-run effects of earthquakes on welfare in rural Indonesia". Journal of Development Economics 118:26–44.
- Guardado, Jenny, and S Pennings. 2021. <u>The seasonality of conflict</u>. World Bank Policy Research WP 9373.
- Harari, Mariaflavia, and Eliana La Ferrara. 2018. "Conflict, climate, and cells: a disaggregated analysis". Review of Economics and Statistics 100 (4): 594–608.

#### Hardeweg, Bernd. 2001.

A conceptual framework for economic evaluation of Desert locust management interventions. Vol. Special Issue No. 5.

- Harris, IPDJ, et al. 2014. "Updated high-resolution grids of monthly climatic observations-the CRU TS3. 10 Dataset". International journal of climatology 34 (3): 623-642.
- Heger, Martin Philipp, and Eric Neumayer. 2019. "The impact of the Indian Ocean tsunami on Aceh's long-term economic growth". Journal of Development Economics 141:102365.
- Hsiang, Solomon M. 2010. "Temperatures and cyclones strongly associated with economic production in the Caribbean and Central America". Proceedings of the National Academy of sciences 107 (35): 15367–15372.
- Hsiang, Solomon M, and Marshall Burke. 2014. "Climate, conflict, and social stability: what does the evidence say?" Climatic change 123 (1): 39–55.

Hsiang, Solomon M, and Amir S Jina. 2014. <u>The causal effect of environmental catastrophe on long-run economic growth: Evidence from 6,700 cyclowed National Bureau of Economic Research WP20352.</u>

- Iyigun, Murat, Nathan Nunn, and Nancy Qian. 2017. <u>The long-run effects of agricultural productivity on conflict, 1400-1900</u>. National Bureau of Economic Research WP24066.
- Joffe, Steen. 2001.
  - Economic and policy issues in desert locust management: A preliminary analysis. FAO/EMPRES Workshop on Economics in Desert Locust Management.

- Klomp, Jeroen, and Kay Valckx. 2014. "Natural disasters and economic growth: A meta-analysis". Global Environmental Change 26:183–195.
- Kocornik-Mina, Adriana, et al. 2020. "Flooded cities". American Economic Journal: Applied Economics 12 (2): 35–66.
- Krall, S, and C Herok. 1997. "Economics of desert locust control". In <u>New strategies in</u> locust control, 401–413. Springer.
- Lazar, Mohammed, et al. 2016. "Importance of solitarious desert locust population dynamics: lessons from historical survey data in A lgeria".Entomologia experimentalis et applicata 161 (3): 168–180.
- Lecoq, Michel. 2001. "Recent progress in Desert and Migratory Locust management in Africa. Are preventative actions possible?" Journal of Orthoptera Research 10 (2): 277–291.
- Liu, Ye, and Yongkang Xue. 2020. "Expansion of the Sahara Desert and shrinking of frozen land of the Arctic". Scientific reports 10 (1): 1–9.
- Maystadt, Jean-François, and Olivier Ecker. 2014. "Extreme weather and civil war: Does drought fuel conflict in Somalia through livestock price shocks?" American Journal of Agricultural Economics 96 (4): 1157–1182.
- McGuirk, Eoin F, and Nathan Nunn. 2021. <u>Transhumant Pastoralism, Climate Change, and Conflict in Africa</u>. National Bureau of Economic Research WP 28243.
- McGuirk, Eoin, and Marshall Burke. 2020. "The economic origins of conflict in Africa". Journal of Political Economy 128 (10): 3940–3997.
- Miguel, Edward, Shanker Satyanath, and Ernest Sergenti. 2004. "Economic shocks and civil conflict: An instrumental variables approach". Journal of political Economy 112 (4): 725–753.
- Oerke, E-C. 2006. "Crop losses to pests". <u>The Journal of Agricultural Science</u> 144 (1): 31–43.
- Rahimi, Jaber, et al. 2021. "Beyond livestock carrying capacity in the Sahelian and Sudanian zones of West Africa". Scientific reports 11 (1): 1–15.
- Raleigh, Clionadh, et al. 2010. "Introducing ACLED: an armed conflict location and event dataset: special data feature". Journal of peace research 47 (5): 651–660.
- Ramankutty, Navin, et al. 2010. "Global Agricultural Lands: Croplands/Pastures, 2000". Http://sedac.ciesin.columbia.edu/es/aglands.html, Socioeconomic Data and Applications Center (SEDAC).
- Renier, Cécile, et al. 2015. "A dynamic vegetation senescence indicator for near-real-time desert locust habitat monitoring with MODIS". Remote Sensing 7 (6): 7545–7570.
- Roy, Andrew Donald. 1951. "Some thoughts on the distribution of earnings". Oxford economic papers 3 (2): 135–146.
- Sarsons, Heather. 2015. "Rainfall and conflict: A cautionary tale". Journal of development Economics 115:62–72.

- Showler, Allan T. 2019. "Desert locust control: the effectiveness of proactive interventions and the goal of outbreak prevention". American Entomologist 65 (3): 180–191.
- Sundberg, Ralph, and Erik Melander. 2013. "Introducing the UCDP georeferenced event dataset". Journal of Peace Research 50 (4): 523–532.
- Symmons, PM, and K Cressman. 2001. "Desert locust guidelines: biology and behaviour". Food and Agriculture Organization of the United Nations.
- The United States Department of Agriculture (USDA). 2022. <u>Crop Calendar Charts</u>. Foreign Agricultural Service, International Production Assessment Division. https://ipad.fas.usda.gov/ogamaps/cropcalendar.aspx.
- The World Bank. 2020. <u>The locust crisis: The World Bank's response</u>. Factsheet. https://www.worldbank.org/en/news/factsheet/2020/04/27/the-locust-crisis-the-world-banks-response.
- Ubilava, David, Justin V Hastings, and Kadir Atalay. 2022. "Agricultural Windfalls and the Seasonality of Political Violence in Africa". American Journal of Agricultural Economics DOI: 10.1111/ajae.12364.
- Witsenburg, Karen M, and Wario R Adano. 2009. "Of rain and raids: Violent livestock raiding in northern Kenya". Civil Wars 11 (4): 514–538.
- Zhang, Long, et al. 2019. "Locust and grasshopper management". <u>Annu. Rev. Entomol</u> 64 (1): 15–34.

# Figures



Figure 1: Desert locust observations by year, study period

Map created by authors using swarms observations retrieved from the FAO Locust Watch database.



Figure 2: Distribution of swarm and violent conflict observations over sample countries

Land used for agriculture includes crop land and pasture land. This panel shows most clearly which countries in West, Central, and East Africa are excluded from the study sample.

Figure 3: Effect of locust swarms at varying distances from the cell



The figure shows point estimates and 95% confidence intervals for estimates of the impact of locust swarms by location relative to the cell on the probability of violent conflict in the cell. Observations are grid cells approximately  $28 \text{km}^2$  by year. All regressions include weather controls as well as country-year and cell FE.

Figure 4: Effects of 2003-2005 locust upsurge on the risk of conflict by year



The dependent variable is a dummy for any violent conflict event observed. Coefficients are for the interaction of a dummy for being in a cell that had any swarm between 2003-2005 with year. The reference year is 2003, indicated by the red line. Bars represent 95% confidence intervals. The regression includes controls for current swarms, weather, and cell and country-by-year FE. Observations are grid cells approximately 28km<sup>2</sup> by year. SEs are clustered at the country level.

# Tables

	(1)	(2)	(3)	(4)	(5)	(6)
	All cells	>=10,000 population	Any crop or pasture land	Ever had a swarm w/in 100km	Ever had a swarm in cell	Ever had a violent conflict in cell
Any swarm in cell	$-0.015^{***}$ (0.005)	$-0.021^{***}$ (0.006)	$-0.019^{***}$ (0.005)	$-0.015^{***}$ (0.005)	$-0.008^{**}$ (0.004)	$-0.024^{***}$ (0.007)
Any swarm in cell previous year	$-0.009^{*}$ (0.004)	-0.005 (0.008)	$-0.009^{*}$ (0.004)	$-0.008^{*}$ (0.004)	-0.002 (0.005)	-0.008 (0.009)
Any swarm within 100km outside cell	-0.001 (0.002)	$\begin{array}{c} 0.003 \\ (0.004) \end{array}$	-0.000 (0.003)	$\begin{array}{c} 0.001 \\ (0.002) \end{array}$	-0.001 (0.004)	$\begin{array}{c} 0.003 \\ (0.009) \end{array}$
Any swarm within 100km outside cell previous year	$\begin{array}{c} 0.004 \\ (0.008) \end{array}$	$\begin{array}{c} 0.001 \\ (0.006) \end{array}$	$0.007 \\ (0.010)$	$\begin{array}{c} 0.006 \\ (0.009) \end{array}$	$\begin{array}{c} 0.001 \\ (0.008) \end{array}$	$\begin{array}{c} 0.002 \\ (0.009) \end{array}$
Total annual rainfall $(100 \text{ mm})$	$\begin{array}{c} 0.003^{*} \\ (0.002) \end{array}$	$\begin{array}{c} 0.002\\ (0.002) \end{array}$	$\begin{array}{c} 0.003 \\ (0.002) \end{array}$	$\begin{array}{c} 0.003 \\ (0.002) \end{array}$	$\begin{array}{c} 0.005 \\ (0.004) \end{array}$	$0.008^{*}$ (0.005)
Total annual rainfall previous year $(100 \text{ mm})$	$\begin{array}{c} 0.003 \\ (0.002) \end{array}$	$\begin{array}{c} 0.003 \\ (0.002) \end{array}$	$\begin{array}{c} 0.003 \\ (0.002) \end{array}$	$0.003^{*}$ (0.002)	$\begin{array}{c} 0.004 \\ (0.003) \end{array}$	$\begin{array}{c} 0.003 \ (0.003) \end{array}$
Max annual temperature (deg C)	$0.006^{*}$ (0.003)	$\begin{array}{c} 0.003 \\ (0.005) \end{array}$	$0.008 \\ (0.006)$	$\begin{array}{c} 0.007 \\ (0.006) \end{array}$	$\begin{array}{c} 0.008 \\ (0.006) \end{array}$	$0.019^{*}$ (0.011)
Max annual temperature previous year (deg C)	$\begin{array}{c} 0.005 \\ (0.004) \end{array}$	$\begin{array}{c} 0.002\\ (0.008) \end{array}$	$\begin{array}{c} 0.007 \\ (0.008) \end{array}$	$0.008 \\ (0.007)$	$\begin{array}{c} 0.006 \\ (0.007) \end{array}$	-0.006 (0.013)
Observations Outcome mean, no swarms Proportional effect of swarms Country-Year FE Cell FE	508284 0.020 -0.755 Yes Yes	148522 0.048 -0.446 Yes Yes	269850 0.034 -0.545 Yes Yes	283214 0.025 -0.577 Yes Yes	50404 0.040 -0.210 Yes Yes	71234 0.142 -0.169 Yes Yes

Table 1:	Effect	of lo	ocust	swarms	on	the	risk	of	conflict,	by	subsample	)

The dependent variable is a dummy for any violent conflict event observed. Observations are grid cells approximately  $28 \text{km}^2$  by year. SEs clustered at the country level are in parentheses. SEs for estimates in column (1) using different clustering approaches are reported in Figure A7. \* p < 0.1, \*\* p < 0.05, \*\*\* p < 0.01

	(1)	(2)	(3)	(4)	(5)	(6)
	$0.25~{\rm deg}$	$0.5~{\rm deg}$	$1 \deg$	$2 \deg$	$5 \deg$	Country
Any swarm in cell	-0.015***	-0.024***	-0.033***	-0.019	0.056	0.015
	(0.005)	(0.006)	(0.009)	(0.020)	(0.036)	(0.027)
Any swarm in cell previous	-0.009*	$-0.011^{*}$	-0.008	-0.013	$0.059^{*}$	-0.062
year	(0.004)	(0.006)	(0.010)	(0.022)	(0.032)	(0.044)
Any swarm within 100km outside	0.004	0.008	0.012	0.031	-0.005	0.026
cell previous year	(0.008)	(0.009)	(0.009)	(0.021)	(0.027)	(0.049)
Total annual rainfall $(100 \text{ mm})$	$0.003^{*}$	$0.005^{**}$	0.003	0.009	0.005	$0.027^{*}$
	(0.002)	(0.002)	(0.004)	(0.005)	(0.013)	(0.015)
Total annual rainfall previous	0.003	0.005	0.004	0.009	-0.006	0.005
year $(100 \text{ mm})$	(0.002)	(0.003)	(0.006)	(0.009)	(0.012)	(0.017)
Max annual temperature (deg C)	$0.006^{*}$	$0.009^{*}$	$0.015^{**}$	0.008	-0.014	-0.013
	(0.003)	(0.005)	(0.006)	(0.007)	(0.010)	(0.030)
Max annual temperature	0.005	0.007	0.003	-0.021***	-0.039***	-0.016
previous year (deg C)	(0.004)	(0.007)	(0.006)	(0.006)	(0.012)	(0.023)
Observations	508284	139342	40823	13312	3673	483
Outcome mean, no swarms	0.020	0.053	0.117	0.214	0.358	0.809
Country-Year FE	Yes	Yes	Yes	Yes	Separate	Separate
Cell FE	Yes	Yes	Yes	Yes	Yes	INO

Table 2: Effect of locust swarms on the risk of conflict at different scales

The dependent variable is a dummy for any violent conflict event observed in the aggregated area in a year. Swarm presence variables are also dummies at the level of the aggregated area in a year. Results using the share of  $0.25^{\circ}$  cells in the aggregated area with any conflict or swarm event in a year are shown in Table A6. Weather controls are means for total annual rainfall and max annual temperature across cell-years within the aggregated area. Column (1) replicates Column (1) from Table 1. Subsequent columns incrementally increase the size of the spatial units in the analysis. Observations are grid cells of particular size (in terms of degrees) in Columns (1) to (5) and countries in Column (6), in a particular year. SEs are clustered at the country level.

\* p < 0.1, \*\* p < 0.05, \*\*\* p < 0.01

	(1) Any crop land	(2) Any pasture land
Any swarm in cell	-0.005 (0.004)	-0.006 (0.004)
Any swarm in cell $\times$ L and	$-0.024^{**}$ (0.010)	$-0.012^{*}$ (0.006)
Any swarm in cell previous year	-0.005 (0.004)	-0.008 (0.005)
Any swarm in cell previous year $\times$ Land	-0.010 (0.006)	-0.002 (0.006)
Any swarm within 100km outside cell	$0.000 \\ (0.001)$	$0.000 \\ (0.002)$
Any swarm within 100km outside cell $\times$ L and	-0.004 (0.004)	-0.003 (0.003)
Any swarm within 100km outside cell previous year	$0.003 \\ (0.005)$	$0.002 \\ (0.003)$
Any swarm within 100km outside cell previous year $\times$ Land	$0.003 \\ (0.010)$	$0.004 \\ (0.008)$
Observations Outcome mean, no swarms and Land=0 Outcome mean, no swarms and Land=1	$508284 \\ 0.009 \\ 0.047$	508284 0.004 0.034
Country-Year FE Cell FE	Yes Yes	Yes Yes

Table 3: Effect of locust swarms on the risk of conflict, by land cover

The dependent variable is a dummy for any violent conflict event observed. Observations are grid cells approximately  $28 \text{km}^2$  by year. SEs are clustered at the country level. \* p < 0.1, \*\* p < 0.05, \*\*\* p < 0.01

	(1)	(2)	(3)
	All	Ag	Crop
	cells	cells	cells
Any off season swarm in cell	$-0.015^{*}$	-0.018	$-0.039^{**}$
	(0.009)	(0.012)	(0.016)
Any planting season swarm in cell	$-0.018^{*}$	-0.017	-0.022
	(0.009)	(0.010)	(0.013)
Any growing season swarm in cell	-0.005	-0.007	-0.008
	(0.004)	(0.005)	(0.007)
Any harvest season swarm in cell	$\begin{array}{c} 0.000 \\ (0.005) \end{array}$	$\begin{array}{c} 0.002 \\ (0.006) \end{array}$	$\begin{array}{c} 0.002 \\ (0.009) \end{array}$
Any swarm in cell previous	$-0.009^{*}$	$-0.009^{*}$	-0.011
year	(0.004)	(0.004)	(0.006)
Any swarm within 100km outside cell	-0.002 (0.002)	-0.001 (0.003)	$\begin{array}{c} 0.000 \\ (0.004) \end{array}$
Any swarm within 100km outside cell previous year	$\begin{array}{c} 0.004 \\ (0.008) \end{array}$	$\begin{array}{c} 0.006 \\ (0.010) \end{array}$	$\begin{array}{c} 0.001 \\ (0.007) \end{array}$
Observations	508284	269850	145448
Outcome mean, no swarms	0.020	0.034	0.047
Weather controls	Yes	Yes	Yes
Country-year FE	Yes	Yes	Yes
Cell FE	Ves	Ves	Ves

Table 4: Effect of locust swarms on the risk of conflict, by swarm timing and land cover

The dependent variable is a dummy for any violent conflict event observed. Locust swarm observations are matched to agricultural activities based on the month of their arrival and country-level crop calendars. Observations are grid cells approximately  $28 \text{km}^2$  by year. SEs are clustered at the country level. \* p < 0.1, \*\* p < 0.05, \*\*\* p < 0.01

Table 5: Long-term effects of 2003-2005 locust upsurge on the risk of conflict

	(1) All cells	(2) Within 250km of swarm during upsurge	(3) Within 100km of swarm during upsurge
Any swarm in cell	-0.009**	-0.005*	-0.003
	(0.004)	(0.002)	(0.003)
Any swarm in cell previous	-0.012**	-0.011*	-0.012
year	(0.005)	(0.007)	(0.007)
Any 2003-05 swarm $\times$ Post	$0.016^{*}$	$0.017^{**}$	0.016**
	(0.008)	(0.008)	(0.007)
Observations	508284	320463	187953
Outcome mean, no 2003-2005 swarms	0.020	0.015	0.017
Outcome mean post-2005, no 2003-2005 swarms	0.026	0.019	0.023
Country-Year FE	Yes	Yes	Yes
Cell FE	Yes	Yes	Yes
Swarm band and weather controls	Yes	Yes	Yes

The dependent variable is a dummy for any violent conflict event observed. Observations are grid cells approximately 28km<sup>2</sup> by year. SEs are clustered at the country level. \* p < 0.1, \*\* p < 0.05, \*\*\* p < 0.01

# Appendix A: Additional Figures and Tables



Figure A1: Example crop calendars

Source: U.S. Department of Agriculture Foreign Agricultural Service, International Produlsion Assessment Division (USDA 2022).

The Libya crop calendar is fairly representative of other North African countries, and the Mali crop calendar is fairly representative of other West African countries.



Figure A2: Timing of locust swarm arrival by phase of crop calendar and region

Agricultural activities by month are determined by assigning each cell with any crop land the primary activity for that month in the country in which it is located, using USDA 2022 crop calendars. Land cover in the year 2000 is from Ramankutty et al. (2010). Locust swarm observations are matched to agricultural activities based on the location and month of their arrival.

	Mean	SD	Min	$25^{th}$	$50^{th}$	$75^{th}$	Max	Ν
Any violent conflict event - ACLED	0.02	0.14	0.0	0.0	0.0	0.0	1.0	513626
Any violent conflict event in cell in any year	0.14	0.35	0.0	0.0	0.0	0.0	1.0	513626
Any swarm in cell	0.01	0.07	0.0	0.0	0.0	0.0	1.0	513626
Any swarm within 100km outside cell	0.04	0.21	0.0	0.0	0.0	0.0	1.0	513626
Any swarm within 100-250km of cell	0.11	0.31	0.0	0.0	0.0	0.0	1.0	513626
Any swarm in cell previous year	0.01	0.07	0.0	0.0	0.0	0.0	1.0	513626
Any swarms within cell in any year	0.10	0.30	0.0	0.0	0.0	0.0	1.0	513626
Any swarms within 100 km in any year	0.56	0.50	0.0	0.0	1.0	1.0	1.0	513626
Population $(10,000s)$	1.75	9.30	0.0	0.0	0.2	1.0	749.8	464708
Total annual rainfall (100 mm)	2.47	3.81	0.0	0.3	0.9	3.0	43.4	508292
Max annual temperature $(\deg C)$	37.55	5.18	12.4	33.8	38.1	41.4	49.0	508292
Any cropland or pasture in cell	0.53	0.50	0.0	0.0	1.0	1.0	1.0	513626
Share of crop and pasture land in cell	0.24	0.32	0.0	0.0	0.0	0.5	1.0	502349
Any cropland in cell	0.28	0.45	0.0	0.0	0.0	1.0	1.0	513626
Share of cropland in cell	0.05	0.13	0.0	0.0	0.0	0.0	1.0	502349
Any pasture in cell	0.52	0.50	0.0	0.0	1.0	1.0	1.0	513626
Share of pasture in cell	0.19	0.27	0.0	0.0	0.0	0.3	1.0	502349

Table A1:	Summary	statistics
-----------	---------	------------

Observations are grid cells approximately  $28 \text{km}^2$  by year.

	(1)	(2)	(3)	(4)	(5) Violent	(6)	(7)
	Locust swarm	Locust swarm AugDec.	Violent conflict	Violent conflict	conflict, no prior yr conflict	Violent conflict AugDec.	Violent conflict AugDec.
Any violent conflict in cell previous year	$-0.005^{*}$ (0.002)			$\begin{array}{c} 0.237^{***} \\ (0.046) \end{array}$			
Any swarm in cell previous year	$-0.026^{***}$ (0.009)	$-0.028^{***}$ (0.003)	$-0.009^{*}$ (0.004)	$-0.006^{*}$ (0.003)	-0.005 (0.003)	$-0.006^{*}$ (0.003)	-0.005 (0.003)
Any swarm within 100km outside cell	$\begin{array}{c} 0.076^{***} \\ (0.008) \end{array}$	$\begin{array}{c} 0.028^{***} \\ (0.007) \end{array}$	-0.001 (0.002)	$-0.002^{**}$ (0.001)	$-0.002^{**}$ (0.001)	-0.002 (0.001)	-0.002 (0.001)
Any swarm within 100km outside cell previous year	$\begin{array}{c} 0.004^{**} \\ (0.001) \end{array}$	$\begin{array}{c} 0.000 \\ (0.001) \end{array}$	$\begin{array}{c} 0.004 \\ (0.008) \end{array}$	$\begin{array}{c} 0.004 \\ (0.007) \end{array}$	$\begin{array}{c} 0.005 \\ (0.006) \end{array}$	$\begin{array}{c} 0.005 \ (0.006) \end{array}$	$\begin{array}{c} 0.004 \\ (0.004) \end{array}$
Total annual rainfall (100 mm)	$\begin{array}{c} 0.000 \\ (0.001) \end{array}$	-0.001 (0.001)	$0.003^{*}$ (0.002)	$0.003^{*}$ (0.001)	$0.002^{**}$ (0.001)	$0.003^{**}$ (0.001)	$0.002^{***}$ (0.001)
Total annual rainfall previous year $(100 \text{ mm})$	-0.000 (0.001)	-0.000 (0.001)	$\begin{array}{c} 0.003 \\ (0.002) \end{array}$	$\begin{array}{c} 0.002 \\ (0.002) \end{array}$	$\begin{array}{c} 0.002 \\ (0.001) \end{array}$	$\begin{array}{c} 0.001 \\ (0.001) \end{array}$	$\begin{array}{c} 0.001 \\ (0.001) \end{array}$
Max annual temperature (deg C)	-0.001 (0.001)	-0.000 (0.001)	$0.006^{*}$ (0.003)	$\begin{array}{c} 0.005^{*} \\ (0.003) \end{array}$	$0.003^{**}$ (0.001)	$\begin{array}{c} 0.004 \\ (0.003) \end{array}$	$\begin{array}{c} 0.002\\ (0.002) \end{array}$
Max annual temperature previous year (deg C)	$\begin{array}{c} 0.002 \\ (0.001) \end{array}$	-0.001 (0.001)	$\begin{array}{c} 0.005 \\ (0.004) \end{array}$	$\begin{array}{c} 0.004 \\ (0.003) \end{array}$	$\begin{array}{c} 0.002\\ (0.002) \end{array}$	$\begin{array}{c} 0.005 \\ (0.003) \end{array}$	$\begin{array}{c} 0.004 \\ (0.002) \end{array}$
Any swarm in Aug-Dec in cell		$\begin{array}{c} 0.107^{*} \\ (0.056) \end{array}$				$-0.005^{*}$ (0.003)	-0.002 (0.002)
Any violent conflict event in Jan-Jul in cell		-0.002 (0.001)					$\begin{array}{c} 0.277^{***} \\ (0.050) \end{array}$
Any swarm in cell			$-0.015^{***}$ (0.005)	$-0.012^{***}$ (0.004)	$-0.008^{***}$ (0.002)		
Observations Outcome mean	508284 0.005	508284 0.003	508284 0.020	508284 0.020	499306 0.011	508284 0.012	508284 0.012

Table A2: Effect of locust swarms on the risk of conflict, controlling for lagged conflict

Columns indicate which dummy dependent variable is used. The first two columns test impacts of prior conflict on the probability of observing a locust swarm. The remaining columns test the impact of locust swarms on the probability of observing violent conflict; column (3) replicates column (1) from Table 1. Observations are grid cells approximately 28km<sup>2</sup> by year. All columns include cell and country-year FEs. SEs clustered at the country level are in parentheses.

\* p < 0.1, \*\* p < 0.05, \*\*\* p < 0.01

Figure A3: Trend in risk of violent conflict over time, by experience of 2003-2005 locust upsurge



The figure shows the share of cells experiencing any violent conflict by year, separately for cells that did and did not experience any locust swarms during the 2003-2005 upsurge. Observations are grid cells approximately 28km<sup>2</sup> by year. SEs are clustered at the country level.

## Appendix B: Desert locusts background

The desert locust is considered the world's most dangerous and destructive migratory pest (Cressman, Van der Elstraeten, and Pedrick 2016; Lazar et al. 2016). Damages from locust shocks can be extreme, with a small swarm covering one square kilometer can consume as much food in one day as 35,000 people. During the last locust upsurge in 2003-2005 in North and West Africa, 100, 90, and 85% losses on cereals, legumes, and pastures respectively were recorded, affecting more than 8 million people (Renier et al. 2015; Brader et al. 2006). Damages to crops alone were estimated at \$2.5 billion USD and \$450 million USD was required to bring an end to the upsurge (ASU 2020).

In the most recent upsurge from 2019-2021 in East Africa and the Arabian Peninsula, over 40 million people in 10 countries faced severe food insecurity due to crop destruction. Locust control operations undertaken by the United Nations Food and Agriculture Organization (FAO) and its partners, primarily via ground and aerial spraying of pesticides, and global food aid efforts helped reduce the damages (Food and Agriculture Organization of the United Nations (FAO) 2022a). The FAO estimates that 3.5 million people were affected by locust destruction, but that control efforts saved agricultural production worth \$1.7 billion USD.

Small numbers of locusts are always present in desert 'recession' areas from Mauritania to India (Figure A4). The population can grow exponentially under favorable climate conditions: periods of repeated rainfall and vegetation growth overlapping with the breeding cycle. The 2019-2021 upsurge persisted in large part because of repeated heavy precipitation out of season, prompting explosive reproduction (Cressman and Ferrand 2021).



Figure A4: Desert locust recession and breeding areas

Source: Symmons and Cressman (2001)

Unique among grasshopper species, after reaching a particular population density desert locusts undergo a process of 'gregarization' wherein they mature physically and form large bands or swarms (Symmons and Cressman 2001). This process typically leads to 'outbreaks,' where locusts spread out from their largely desert initial breeding areas. Locusts in swarms have increased appetites and accelerated reproductive cycles, and are particularly threatening to agriculture because they fly in search of ideal feeding and reproduction location. The FAO distinguishes different levels of locust activity (Symmons and Cressman 2001). We use the terms 'outbreak' and 'upsurge' interchangeably to refer to any locust swarm activity. By the FAO definition 'outbreaks' refer to localized increases in locust numbers while 'upsurges' refer to broader and more sustained locust activities. A third level, 'plagues,' is characterized by larger and more widespead locust infestations.

As illustrated by Figure A5, locust swarms are not observed with any regularity over time or space. Desert locusts are migratory, moving on after consuming all available vegetation, and outside of outbreak periods are ultimately restricted to desert 'recession' areas. Unlike many other insect species, therefore, the arrival of a desert locust swarm does not signal a permanent change in local agricultual pest risk. Instead, the arrival of a swarm can be considered a locally and temporally concentrated natural disaster where all crops and pastureland are at risk (Hardeweg 2001).



2020



Map created by authors using swarms observations retrieved from the FAO Locust Watch database.

The frequency of large-scale outbreaks has fallen since around the 1980s, in large part due to increases in coordinated preventive operations (Cressman and Stefanski 2016), as shown by the figure below. Given their tolerance for extreme heat and responsiveness to periods of heavy precipitation, however, climate change might create conditions conducive to more frequent desert locust outbreaks.

Locust outbreaks end due to a combination of migration to unfavorable habitats, failure of seasonal rains, and control operations (Symmons and Cressman 2001). The only current viable method of swarm control is direct air or ground spraying with pesticides (Cressman and Ferrand 2021). These control operations do not prevent immediate agricultural destruction as they take some time to kill the targeted locusts, but will limit their spread. Farmers have no proven effective recourse when faced with the arrival of a locust swarm; some activities



Source: Cressman and Stefanski (2016), Figure 6.

such as setting fires or placing nets on crops may slow damage but have little effect on locust population (Dobson 2001; Hardeweg 2001).

Desert locust control is therefore most effective before locust populations surge, and the FAO manages an international network of early monitoring, warning, and prevention systems in support of this goal (Zhang et al. 2019). While improvements in desert locust management have been largely effective in reducing the frequency of outbreaks, many challenges remain. Desert locust breeding areas are widespread and often in remote or insecure areas. Small breeding groups are easy to miss by monitors, and swarms can migrate quickly. In addition, control operations are slow and costly, resources for monitoring and control are limited outside of upsurges, and the cross-country nature of the thread creats coordination issues.

Locust swarms vary in their density and extent (Symmons and Cressman 2001). The average swarm includes around 50 locusts per  $m^2$  with a range from 20-150, and can cover under  $1km^2$  to several hundred. It is therefore not uncommon for swarms to include over a billion locusts. Swarms fly downwind from a few hours after sunrise to an hour or so before sunset. Seasonal changes in winds tend to bring locusts to an area when rain is most likely, allowing them to continue breeding. The localized nature of locust swarm shocks stems from these patterns of swarm movements. After taking off in the morning, swarms fly downwind for 9-10 hours rather than landing as soon as they encounter new vegetation. A swarm can easily move 100km or more in a day even with minimal wind (Symmons and Cressman 2001). Consequently, the flight path of a locust swarm will include both affected and unaffected areas, with the affected areas determined by largely by patterns of wind direction and speed over time from the initial swarm formation in breeding areas.

## Appendix C: Simple model

A simple stylized model formalizing how changes to returns and opportunity costs of fighting affect conflict risk can be used to generate hypotheses about the effect of agricultural shocks on conflict. The conflict modeled here should be thought of as conflict initiated by non-state actors, as in much of the literature on agriculture shocks and conflict (Crost et al. 2018; Guardado and Pennings 2021; McGuirk and Nunn 2021; Ubilava, Hastings, and Atalay 2022). For simplicity, I focus on a static, partial equilibrium decision about time allocation with two parties.<sup>15</sup>

Consider a version of a simple Roy model (Roy 1951; French and Taber 2011) where two individuals<sup>16</sup> *i* and *j* choose their occupations to maximize net returns in a given time period. In this case, individual *i* decides between household agricultural work with net returns  $F(S_i, X_i)$ , work outside the household farm with net returns (wages)  $w(X_i)$ , and fighting to capture output or factors of production with net returns  $R(X_i, S_j, X_j)$ , where *j* indexes the other party. Net returns are a function of individual and location factors  $X^{17}$ . Returns to agricultural production and fighting also depend on the realization of agricultural production shocks *S*, modeled as continuous between 0 and 1 with S = 0 indicating no shock and S = 1 indicating a shock that results in 0 agricultural production.<sup>18</sup> Non-farm returns do not depend on S.<sup>19</sup>

The returns to fighting  $R(X_i, S_j, X_j)$  are the difference between expected benefits and expected costs. Benefits to *i*—who initiates the fighting—depend on the value of production outputs  $F(S_j, X_j)$  and  $w(X_j)$  and of factors of production in the held by the counterparty *j*, included in  $X_j$ . Factors of production, are valued based on the expected future stream of benefits from their use, bringing a long-term perspective into the static decision. The impact of agricultural shock  $S_j$  is through a reduction in the value of agricultural output that may be captured. Benefits of initiating fighting are received with some probability  $\pi$  of success which depends on  $X_i$  and  $X_j$ . For example, an increase in *i*'s ability and resources relative to *j* increase  $\pi$ . Costs of fighting are incurred with certainty, and include both resource costs as well as potential social and emotional costs. If *i* decides to fight *j*, *j* also incurs some costs regardless of their own occupation decision. I abstract away from the possibility that *j* might capture resources from *i* following *i*'s decision to initiate fighting.

 $<sup>^{15}</sup>$ The model sets aside dynamic considerations in the decision to fight (which Chassang, Miquel, et al. (2009) show to be important) for simplicity, but we can think of the present returns to fighting in the model as incorporating long-term costs and benefits of fighting. We can also extrapolate to multiple parties, but I focus on the case of two for simplicity.

<sup>&</sup>lt;sup>16</sup>We could also conceptualize the unit as representing households, communities, or other groups at which decisions to engage in conflict are made.

 $<sup>^{17}</sup>$ For example, among many potential factors included in X wages will depend on education and urban setting, agricultural production will depend on farm assets (e.g., land, tools), soil characteristics, and weather, and returns to fighting will depend on individual ability, counterparty ability, and value of counterparty output.

<sup>&</sup>lt;sup>18</sup>As I focus on the impact of locust swarms, I abstract away from the possibility of positive agricultural production shocks.

<sup>&</sup>lt;sup>19</sup>Unlike shocks such as droughts and floods which affect agricultural production but may also affect economies and society more broadly, the literature on desert locusts indicates they have limited economic impacts outside of agriculture. The model could easily be generalized to allow non-farm wages to depend on agricultural shock with different effects for different types of agricultural shocks.

The individual's problem is to maximize returns over the choice of work sector Ag and decision to fight D

$$\max_{D,Ag} \left( \left[ F(S_i, X_i)(Ag_i) + w(X_i)(1 - Ag_i) \right] \cdot (1 - D_i) + R(X_i, S_j, X_j) \cdot D_i \right)$$
  
subject to  
$$\frac{\partial F(X_i)}{\partial S_i} < 0; \quad \frac{\partial R(X_i, X_j)}{\partial S_j} < 0$$

For simplicitly and intuition I ignore uncertainty in returns. Suppose that decisions are made (or, equivalently, that they may be changed) after the agricultural shocks  $S_i, S_j$  are realized. In this scenario, the individual will choose the sector with the highest returns and will choose to fight only if the returns to fighting exceed the returns to working, or

$$Ag = 1 \text{ iff } F(S_i, X_i) > w(X_i)$$
$$D = 1 \text{ iff } R(X_i, S_j, X_j) > \max(F(S_i, X_i), w(X_i))$$

Since most individuals do not choose to fight in most time periods, even when the returns to working may be low as in the off season for poor smallholder farm households, I assume that  $R(X_i, 0, X_j)$  is less than  $F(0_i)$  or  $w(X_i)$  for most of the support of  $X_i$ , for example due to factors such as a low probability of success and high economic and social costs to fighting relative to the benefits.

Shock to agricultural production  $S_i$  and  $S_j$  will affect the decision to fight D only if they lead to a change in the inequality. In the case where fighting was not optimal with no agricultural shock, the opportunity cost of fighting—the returns to working in either sector must fall below the returns to fighting. Even if  $\frac{\partial F(X_i)}{\partial S_i}$  is large, returns to non-agricultural work  $w(X_i)$  set a floor on how low the opportunity cost of fighting can fall following a shock  $S_i$ .

This model is ambiguous on the sign of how an agricultural shock will affect the risk of conflict in a particular area. A decrease in the returns to fighting in an area with a negative agricultural shock may deter parties from outside the area from attacking if agricultural output is a large share of the benefit from fighting. But decreased agricultural production in the area may push individuals to fight other parties nearby, even if those parties are also affected by the shock.

Opportunity costs of fighting will be lower on average following an agricultural shock, even if the shock induces some individuals to switch out of agriculture. I assume that  $\frac{\partial w(X_i)}{\partial S_i} = 0$ , but in the case this does not hold it is most likely that  $\frac{\partial w(X_i)}{\partial S_i} < 0$ , which would further decrease the opportunity costs of fighting. The returns from conflict  $R(X_i, S_j, X_j)$ may also fall when  $S_i$  is larger if  $S_i$  and  $S_j$  are spatially correlated, as is the case for most weather shocks, by the decrease in agricultural output available to capture. This decrease will be limited by returns that can be obtained from the capture of factors of production and of non-agricultural output. If the fall in returns to fighting is not be as large as the fall in opportunity cost, this would imply that for some  $X_i$ —and more likely in agricultural areas when  $w(X_i)$  is lower—we will have  $\frac{\partial D}{\partial S} > 0$ . **Hypothesis 1** is thus that agricultural shocks have a positive effect on the likelihood of conflict, on average. Rejecting this hypothesis would imply either that opportunity costs of fighting do not fall as much as expected following a shock, or that the returns to fighting fall by more than expected.

This hypothesis may be more likely to hold with locust shocks since all local vegetation is at risk of being consumed by a locust swarm, meaning  $F(S_i, X_i)$  is close to 0 dramatically decreasing opportunity costs related to agriculture when  $S_i$  is a locust swarm. Locusts swarms also have limited other impacts which could reduce the returns to conflict, such as destruction of property as with floods.

An important characteristics of locust swarms is that the effect on agricultural production will depend on the timing of the swarm, with the largest effects between planting and harvest when crops are growing. Swarms should have limited effects once crops have been harvested and before they are planted, and smaller effects when crops have been planted but have not yet sprouted or could potentially be re-planted.<sup>20</sup> An exception might be if swarms arriving in these seasons are taken as signaling increased risk of additional swarms, which might reduce expected returns to agriculture. The destruction of agricultural output will be most catastrophic if a swarm arrives during the growing season or the start of harvest. At that point, inputs have been committed and after crops have been destroyed the returns to continuing work in agriculture for the year are nearly zero, reducing the opportunity cost of fighting. This is also the lean season when farmers have less food and cash, reducing the resources available to seek non-farm work.

We can formalize these differences by writing  $0 = S_{offseason} < S_{planting} < S_{growing} < S_{earlyharvest} \approx 1$  for locust shocks, where the subscript indicates the timing of the swarm's arrival. Since  $\frac{\partial F(X)}{\partial S} < 0$ , this has implications for the magnitude of the production shock by swarm timing relative to the agricultural cycle. Based on these differences, if Hypothesis 1 holds then the impact of locusts on conflict should be greatest for swarms arriving during the growing season or the start of harvest when they reduce the opportunity cost related to agricultural work the most. A corrolary is that swarms should have little effect if they arrive in the off season. Rejecting this hypothesis would imply that changes in agricultural productivity alone do not explain the impacts of locust swarms on conflict.

While Hypothesis 1 is that the risk of conflict will increase with an agricultural shock, another consideration is *where* conflicts will take place. If i and j are not located in the same location, the model indicates that i's decision to fight leads to conflict in j's location, rather than i's own location. Indeed, Harari and La Ferrara (2018) show that a cell experiencing an agriculture-relevant weather shock increases the risk of conflict in neighboring cells in the same year. **Hypothesis 2** is that agricultural shocks in a particular area, having lowered the opportunity cost of fighting for individuals from that area, will increase the risk of conflict in surrounding areas as neighbors are attacked. Moreover, other parties deciding to fight would be more likely to attack a nearby unaffected area rather than a similar shock-affected area due to the different in agricultural output to capture. These conflict spillovers could lead to decreases in conflict observed in shock-affected areas, even if parties from those areas decide to engage in conflict.

Spillovers could be particularly important for desert locust swarms. Though there is spa-

<sup>&</sup>lt;sup>20</sup>This assumes that annual crops such as staple grains make up the large majority of the value of agricultural production relative to perennial or tree crops and livestock. This is true for much of the sample area, but to the extent it is not that would attenuate any seasonality in the impact of swarms.

tial correlation in locust swarm events,  $S_i > 0$  does not necessarily imply  $S_j > 0$  for locations j near the individual at location i when S represents locust shock due to swarm movement patterns. Locusts often fly large distances between landings, creating more local variation in effects than other agricultural production shocks that typically are highly spatially correlated. This characteristic increases the likelihood that a decrease in  $F(S_i, X_i)$  following a locust shock  $S_i$  reduces the opportunity cost of fighting sufficiently relative to  $R(X_i, S_j, X_j)$  to make fighting optimal for i, creating an ideal setting for evaluating conflict spillovers.

An alternative possibility not captured in the model is that if  $S_i$  is large but  $S_j = 0$  or is small, j could make a transfer to i conditional on i not attacking, in the same way that aid and relief often arrives in communities following agricultural shocks. Such a transfer could prevent conflict if it is larger than the returns to fighting for i and could be rational for j if it is below the cost it would incur if i attacked.

Finally, agricultural shocks may also have long-term impacts on the risk of conflict, though evidence of persistence more than one year after a shock is limited (Crost et al. 2018; Harari and La Ferrara 2018).Short-term impacts operate through the immediate agricultural productivity shock, but agricultural shocks might also affect long-term productivity. The model allows for long-term effects if we think of agricultural assets A for example as one of the X variables affecting agricultural production F(S, X), and make current assets a function of past agricultural shocks. In the simplest sense, we would thus have  $F(S_t, A_t(S_{t-s}), X_t)$ , with  $\frac{\partial F(S_t, X_t)}{\partial A_t} > 0$  and  $\frac{\partial A_t}{\partial S_{t-s}} < 0$ . Following this argument, **Hypothesis 3** is that impacts of agricultural shocks on conflict in an area persist in the long term but decrease over time as affected areas recover. Rejecting this hypothesis would imply that agricultural shocks do not have a persistent effect on agricultural productivity large enough to make fighting optimal. Another possibility is simply that affected individuals do not stay in the same area, breaking the connection between the shock and local conflict.

The arrival of a locust swarm does not increase the likelihood of locust damages in future years or otherwise change local agricultural fundamentals. In addition, locust damages are concentrated on vegetation, unlike shocks such as floods which might affect future productivity by damaging production infrastructure. On the other hand, the damages to agricultural production from locust swarms are severe, and individual efforts to cope with and recover from the locust shock might permanently affect their agricultural productivity. In particular, households that sell assets or send members away to cope with short-term livelihood and food security issues—common coping strategies—could end up with a persistently lower stock of productive assets, reducing productivity and lowering the opportunity cost to fighting. Given the catastrophic nature of locust destruction, the impact on the individual's assets  $\left|\frac{\partial A_t}{\partial S_{t-s}}\right|$  might be particularly large, thus resulting in meaningful reductions in the opportunity cost of fighting related to agricultural production in time periods following a locust swarm's arrival. Since this is paired with no (or limited<sup>21</sup>) reduction in the current returns to fighting from past locust swarms, this would make affected areas more likely to decide to engage in conflict and more vulnerable to conflict following future agricultural shocks.

 $<sup>^{21}</sup>$ Future returns to fighting would fall to the extent agricultural production is also persistently lower following a past swarm.

## Appendix D: Robustness



Figure A7: Estimated coefficients from Equation 1 with different SEs

The figure shows 95% confidence intervals for estimates from Table 1 column (1) applying different clustering for the SEs. The outcome variable is a dummy for any violent conflict observed. Observations are grid cells approximately  $28 \text{km}^2$  by year. Regressions also include country-year and cell FE.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)
Any swarm in cell	-0.009 (0.008)	-0.010 (0.006)	-0.009 (0.008)	$-0.015^{***}$ (0.005)	$-0.015^{***}$ (0.005)	$-0.013^{***}$ (0.004)	$-0.013^{**}$ (0.006)
Any swarm within 100km outside cell		$\begin{array}{c} 0.001 \\ (0.004) \end{array}$			-0.001 (0.002)	-0.001 (0.001)	-0.003 (0.002)
Any swarm in cell previous year			-0.008 (0.005)		$-0.009^{*}$ (0.004)	$-0.007^{*}$ (0.004)	-0.008 (0.006)
Any swarm within 100km outside cell previous year					$\begin{array}{c} 0.004 \\ (0.008) \end{array}$	-0.000 (0.002)	$\begin{array}{c} 0.008 \\ (0.009) \end{array}$
Observations	562539	562539	562539	508284	508284	507922	508284
$Adj-R^2$	0.316	0.316	0.316	0.320	0.320	0.385	0.275
Outcome mean, no swarms	0.020	0.020	0.020	0.020	0.020	0.020	0.020
Proportional effect of swarms	-0.449	-0.485	-0.455	-0.773	-0.755	-0.685	-0.659
Swarm Bands	No	Yes	No	No	Yes	Yes	Yes
Swarm Lags	No	No	Yes	No	Yes	Yes	Yes
Weather	No	No	No	Yes	Yes	Yes	Yes
Time FE	Cntry-year	Cntry-year	Cntry-year	Cntry-year	Cntry-year	Region-year	Year
Location FE	Cell	Cell	Cell	Cell	Cell	Cell	Cell

Table A3: Effect of swarms on the risk of conflict, varying controls

The outcome variable is a dummy for any violent conflict observed. The main independent variable is a dummy for any locust swarms observed. Observations are grid cells approximately  $28 \text{km}^2$  by year. SEs are clustered at the country level.

\* p < 0.1, \*\* p < 0.05, \*\*\* p < 0.01

Table A4: Effect of swarms on the risk of conflict, omitting particular years and regions

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Any swarm in cell	$-0.015^{***}$ (0.005)	$-0.010^{***}$ (0.003)	$-0.007^{**}$ (0.003)	-0.023 (0.026)	$-0.013^{**}$ (0.006)	$-0.016^{**}$ (0.006)	$-0.018^{**}$ (0.006)	$-0.013^{***}$ (0.004)
Any swarm in cell previous year	$-0.009^{*}$ (0.004)	-0.004 (0.003)	$\begin{array}{c} 0.000\\ (0.002) \end{array}$	-0.011 (0.011)	-0.006 (0.004)	$-0.012^{**}$ (0.005)	-0.009 (0.006)	-0.007 (0.005)
Any swarm within 100km outside cell	-0.001 (0.002)	$-0.003^{**}$ (0.001)	$\begin{array}{c} 0.001 \\ (0.002) \end{array}$	-0.017 (0.012)	-0.001 (0.002)	-0.002 (0.002)	-0.000 (0.002)	-0.002 (0.002)
Any swarm within 100km outside cell previous year	$\begin{array}{c} 0.004 \\ (0.008) \end{array}$	$\begin{array}{c} 0.003 \\ (0.007) \end{array}$	-0.000 (0.001)	$\begin{array}{c} 0.009 \\ (0.013) \end{array}$	$\begin{array}{c} 0.006\\ (0.010) \end{array}$	$ \begin{array}{c} 0.009 \\ (0.010) \end{array} $	$ \begin{array}{c} 0.006 \\ (0.011) \end{array} $	$-0.004^{*}$ (0.002)
Total annual rainfall (100 mm)	$\begin{array}{c} 0.003^{*} \\ (0.002) \end{array}$	$\begin{array}{c} 0.002^{*} \\ (0.001) \end{array}$	$\begin{array}{c} 0.000 \\ (0.001) \end{array}$	$\begin{array}{c} 0.002 \\ (0.003) \end{array}$	$ \begin{array}{c} 0.004 \\ (0.002) \end{array} $	$\begin{array}{c} 0.001 \\ (0.001) \end{array}$	$ \begin{array}{c} 0.004 \\ (0.003) \end{array} $	$\begin{array}{c} 0.004^{*} \\ (0.002) \end{array}$
Total annual rainfall previous year (100 mm)	$\begin{array}{c} 0.003 \\ (0.002) \end{array}$	$\begin{array}{c} 0.002 \\ (0.002) \end{array}$	$\begin{array}{c} 0.002 \\ (0.001) \end{array}$	$\begin{array}{c} 0.002 \\ (0.003) \end{array}$	$0.004^{**}$ (0.002)	$ \begin{array}{c} 0.000 \\ (0.002) \end{array} $	$ \begin{array}{c} 0.002 \\ (0.002) \end{array} $	$\begin{array}{c} 0.003 \\ (0.002) \end{array}$
Max annual temperature (deg C)	$\begin{array}{c} 0.006^{*} \\ (0.003) \end{array}$	$\begin{array}{c} 0.005^{**} \\ (0.002) \end{array}$	$\begin{array}{c} 0.004 \\ (0.003) \end{array}$	$\begin{array}{c} 0.003 \\ (0.006) \end{array}$	$0.009 \\ (0.006)$	$ \begin{array}{c} 0.005 \\ (0.004) \end{array} $	$ \begin{array}{c} 0.008 \\ (0.005) \end{array} $	$\begin{array}{c} 0.004^{*} \\ (0.002) \end{array}$
Max annual temperature previous year (deg C)	$\begin{array}{c} 0.005 \\ (0.004) \end{array}$	$\begin{array}{c} 0.004 \\ (0.003) \end{array}$	$\begin{array}{c} 0.002 \\ (0.002) \end{array}$	$\begin{array}{c} 0.004 \\ (0.008) \end{array}$	$ \begin{array}{c} 0.009 \\ (0.007) \end{array} $	$ \begin{array}{c} 0.005 \\ (0.005) \end{array} $	$ \begin{array}{c} 0.006 \\ (0.006) \end{array} $	$\begin{array}{c} 0.002\\ (0.003) \end{array}$
Observations Outcome mean, no swarms Proportional effect of swarms Regions Years	508284 0.020 -0.755 All All	435679 0.015 -0.637 All 1998-2015	314663 0.010 -0.627 All 1998-2010	217842 0.011 -0.613 All 2010-2018	348010 0.022 -0.598 No N. Africa All	399918 0.012 -1.295 No E. Africa All	350581 0.024 -0.736 No W. Africa All	426342 0.021 -0.613 No Arabia All

The dependent variable is a dummy for any violent conflict event observed. Column (1) replicates column (1) from Table 1. Columns 2-5 restrict the sample of to different time periods. Columns 6-8 restrict the sample of geographies by dropping selected regions from the analysis. Observations are grid cells approximately  $28 \text{km}^2$  by year. All columns include cell and country-year FEs. SEs clustered at the country level are in parentheses.

\* p < 0.1, \*\* p < 0.05, \*\*\* p < 0.01



Figure A8: Impacts of one vs. multiple swarms in a year on violent conflict risk

The figure shows 95% confidence intervals for estimates of the main regressions specification replacing the swarms dummy with dummy variables for either 1 swarm observed in a year or more than 1 swarm, across different outcomes and specifications. The outcome variables are all dummy variables for either ACLED or UCDP violent conflict or for ACLED protests/riots. The legend also indicates what subset of cells is considered for the analysis. Observations are  $0.25^{\circ}$  grid cells—approximately  $28 \text{km}^2$ —by year except where  $0.5^{\circ}$  cells are specified. Regressions also include controls for swarm lags and bands, for weather, and country-year and cell FE.

		Control Mean	Any swarm
	Ν	(SD)	(SE)
Any violent conflict event - ACLED	508284	0.020	-0.015***
		(0.139)	(0.005)
Any violent conflict event - UCDP	508284	0.010	-0.003*
		(0.100)	(0.002)
Any violent state conflict - ACLED	508284	0.013	-0.011***
		(0.114)	(0.004)
Any violent state conflict - UCDP	508284	0.007	-0.001
		(0.086)	(0.002)
Any violent non-state conflict - ACLED	508284	0.013	-0.010***
		(0.112)	(0.004)
Any violent state conflict - UCDP	508284	0.002	0.000
		(0.042)	(0.001)
Any violent one-sided conflict - ACLED	508284	0.011	-0.010***
		(0.106)	(0.004)
Any violent one-sided conflict - UCDP	508284	0.002	-0.001
		(0.049)	(0.001)
Any protest or riot event - ACLED	508284	0.010	-0.008**
		(0.098)	(0.004)
Total fatalities - ACLED	508284	0.695	$-1.000^{*}$
		(52.139)	(0.511)
Total fatalities - UCDP	508284	0.475	-1.772
		(83.310)	(1.807)

Table A5: Effect of swarms on different conflict outcomes

This table shows estimates of the impact of a locust swarm in the same year from Equation 1 using different measures of conflict as the outcome variable. All regressions include nearby swarm and weather controls along with country-year and cell FE. Observations are grid cells approximately 28km<sup>2</sup> by year. SEs are clustered at the country level. \* p < 0.1, \*\* p < 0.05, \*\*\* p < 0.01

\_

	(1)	(2)	(3)	(4)	(5)	(6)
	0.25 deg	0.5 deg	1 deg	$2 \deg$	5 deg	Country
Any swarm in cell	$-0.015^{***}$ (0.005)	$-0.017^{**}$ (0.007)	$-0.018^{*}$ (0.010)	$-0.029^{*}$ (0.015)	-0.013 (0.018)	-0.002 (0.037)
Any swarm in cell previous year	$-0.009^{*}$ (0.004)	-0.010 (0.008)	-0.016 (0.019)	$-0.048^{*}$ (0.028)	-0.074 (0.046)	-0.020 (0.044)
Any swarm within 100km outside cell	-0.001 (0.002)					
Any swarm within 100km outside cell previous year	$0.004 \\ (0.008)$	$0.007 \\ (0.009)$	$0.009 \\ (0.010)$	$\begin{array}{c} 0.015 \\ (0.012) \end{array}$	$0.025 \\ (0.016)$	$0.016 \\ (0.016)$
Total annual rainfall (100 mm)	$0.003^{*}$ (0.002)	$0.003 \\ (0.002)$	$\begin{array}{c} 0.003 \\ (0.002) \end{array}$	$\begin{array}{c} 0.002\\ (0.002) \end{array}$	$0.004^{*}$ (0.002)	$0.002 \\ (0.004)$
Total annual rainfall previous year (100 mm)	0.003 (0.002)	$0.002 \\ (0.002)$	$\begin{array}{c} 0.002 \\ (0.002) \end{array}$	0.003 (0.002)	$\begin{array}{c} 0.001 \\ (0.002) \end{array}$	-0.001 (0.004)
Max annual temperature (deg C)	$0.006^{*}$ (0.003)	$0.005^{*}$ (0.003)	$0.003^{**}$ (0.002)	-0.001 (0.001)	-0.000 (0.001)	-0.006 (0.004)
Max annual temperature previous year (deg C)	$0.005 \\ (0.004)$	$0.004 \\ (0.004)$	$\begin{array}{c} 0.002 \\ (0.002) \end{array}$	-0.001 (0.001)	$-0.002^{*}$ (0.001)	-0.006 (0.003)
Observations	508284	139342	40823	13312	3673	483
Outcome mean, no swarms	0.020	0.020	0.022	0.023	0.023	0.032
Country-Year FE	Yes	Yes	Yes	Yes	Separate	Separate
Cell FE	Yes	Yes	Yes	Yes	Yes	No

Table A6:	Effect	of swarms	on	the	$\operatorname{risk}$	of	$\operatorname{conflict}$	$\operatorname{at}$	$\operatorname{different}$	scales,	taking	means	over
swarm and	conflic	t events											

The dependent variable is the share of 0.25 degree cells in the aggregated area with any violent conflict event in a year. Similarly, swarm presence variables are the share of 0.25 degree cells in the aggregated area with any swarm in a year. Weather controls are means for total annual rainfall and max annual temperature across cell-years within the aggregated area. Column (1) replicates Column (1) from Table 1. Subsequent columns incrementally increase the size of the spatial units in the analysis. Observations are grid cells of particular size (in terms of degrees) in Columns (1) to (5) and countries in Column (6), in a particular year. SEs are clustered at the country level. \* p < 0.1, \*\* p < 0.05, \*\*\* p < 0.01



Figure A9: Effect of swarms on the risk of conflict by season and land cover, by cell size

The figure shows 95% confidence intervals for estimates from Table 4 by land cover in 2000. The outcome variable is a dummy for any violent conflict observed. Observations are grid cells by year, where in panel A we consider the baseline sample of  $0.25^{\circ}$  cells—approximately  $28 \text{km}^2$ —and in panel B we consider  $0.5^{\circ}$  cells—approximately  $56 \text{km}^2$ . Regressions also include controls for locusts in surrounding areas and weather as well as country-year and cell FE.

Figure A10: Seasonal effect of swarms on the risk of conflict, by region and prior year



The figure shows 95% confidence intervals for estimates from Table 4 by region (Panel A) and for swarms the prior year by land cover in 2000 (Panel B). The outcome variable is a dummy for any violent conflict observed. Observations are grid cells approximately 28km<sup>2</sup> by year. Regressions also include controls for locusts in surrounding areas and weather as well as country-year and cell FE.



Figure A11: Effect of 2003-2005 upsurge on the risk of conflict by year in subsamples of cells



A) Cells within 250km of an upsurge swarm

B) Agricultural cells within 250km of an upsurge swarm

The dependent variable is a dummy for any violent conflict event observed. Coefficients are for the interaction of a dummy for being in a cell that had any swarm between 2003-2005 with year. The reference year is 2003, indicated by the red line. Bars represent 95% confidence intervals. The regression includes controls for current swarms, weather, and cell and country-by-year FE. Observations are grid cells approximately  $28 \text{km}^2$  by year. SEs are clustered at the country level.

Figure A12: Effect of 2003-2005 upsurge on different definitions of conflict by year



The dependent variable is a dummy for any conflict event observed, where the definition of conflict in indicated in the panel. Coefficients are for the interaction of a dummy for being in a cell that had any swarm between 2003-2005 with year. The reference year is 2003, indicated by the red line. Bars represent 95% confidence intervals. The regression includes controls for current swarms, weather, and cell and country-by-year FE. Observations are grid cells approximately  $28 \text{km}^2$  by year. SEs are clustered at the country level.